Contents lists available at ScienceDirect

Information Fusion

journal homepage: www.elsevier.com/locate/inffus

A systematic review on multi-criteria group decision-making methods based on weights: Analysis and classification scheme

David Boix-Cots^a, Francesc Pardo-Bosch^{b,c}, Pablo Pujadas^{b,c,*}

^a Department of Civil and Environmental Engineering, Universitat Politècnica de Catalunya BarcelonaTech (UPC), C/ Jordi Girona 1-3, Barcelona 08034, Spain ^b Department of Project and Construction Engineering, Universitat Politècnica de Catalunya BarcelonaTech (UPC), Av. Diagonal 647, Barcelona 08028, Spain

^c Group of Construction Research and Innovation (GRIC), C/ Colom, 11, Ed. TR5, Terrassa, Barcelona 08222, Spain

ARTICLE INFO

Keywords: Group decision-making Multiple criteria group decision-making Expert weights MCGDM MAGDM

ABSTRACT

Interest in group decision-making (GDM) has been increasing prominently over the last decade. Access to global databases, sophisticated sensors which can obtain multiple inputs or complex problems requiring opinions from several experts have driven interest in data aggregation. Consequently, the field has been widely studied from several viewpoints and multiple approaches have been proposed. Nevertheless, there is a lack of general framework. Moreover, this problem is exacerbated in the case of experts' weighting methods, one of the most widely-used techniques to deal with multiple source aggregation. This lack of general classification scheme, or a guide to assist expert knowledge, leads to ambiguity or misreading for readers, who may be overwhelmed by the large amount of unclassified information currently available. To invert this situation, a general GDM framework is presented which divides and classifies all data aggregation techniques, focusing on and expanding the classification of experts' weighting methods in terms of analysis type by carrying out an in-depth literature review. Results are not only classified but analysed and discussed regarding multiple characteristics, such as MCDMs in which they are applied, type of data used, ideal solutions considered or when they are applied. Furthermore, general requirements supplement this analysis such as initial influence, or component division considerations. As a result, this paper provides not only a general classification scheme and a detailed analysis of experts' weighting methods but also a road map for researchers working on GDM topics or a guide for experts who use these methods. Furthermore, six significant contributions for future research pathways are provided in the conclusions.

1. Introduction

Decision-making is a key concept in our everyday lives. Albeit unconsciously, people are constantly choosing the most appropriate option from a set of possible alternatives regarding pre-defined requirements, criteria, or indicators. Consequently, multiple experts have developed structured and scientific decision-making methods rather than intuitive methods. These techniques are known as multiple-criteria decisionmaking (MCDM) methods as they integrate several criteria. They attempt to ensure objective and deliberate analysis to manage challenges successfully [1].

However, constant scientific development and increasingly specific expert knowledge have made problems more complex. Nowadays, a single data source or decision maker (DM) may not be able to consider all relevant aspects of a decision [2] or may not have all the required information. Multiple data sources or decision makers (DMs) could be required for a group decision-making (GDM) process which involves multiple criteria. In these cases, the process is known as multiple-criteria group decision-making (MCGDM) methods. As a consequence, there is a pressing need to explore how to join, incorporate, or consider their opinions to reach a consensus among them and this is constantly discussed among experts [3].

Some 1950s models proposed group negotiation as a solution [4] while others studied the application of election theory [5], which led to the scheme of social choice according to individual values [6]. At the same time, group behavior regarding solution gains and losses was being analysed [7,8]. These studies contributed to the final social choice theory scheme [9].

Besides these psychological aggregation proposals, more mathematical methods arose such as central tendency studies focused on

https://doi.org/10.1016/j.inffus.2023.03.004

Received 24 December 2022; Received in revised form 1 March 2023; Accepted 2 March 2023 Available online 7 March 2023





^{*} Corresponding author at: Department of Civil and Environmental Engineering, Universitat Politècnica de Catalunya · BarcelonaTech (UPC), C/ Jordi Girona 1-3, Barcelona 08034, Spain.

E-mail address: pablo.pujadas@upc.edu (P. Pujadas).

^{1566-2535/© 2023} The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



Fig. 1. Second proposed classification. Source: [23].

finding the statistical value which best fits all DMs [10-12]. Others use the individual DM rankings to aggregate them into a group ranking. The combination techniques include probability distributions [13], Borda heuristic models and their extensions [14,15] or stochastic optimization [16-18].

However, some researchers state that it is impossible to have a homogeneous expert group whose experiences, attitudes, knowledge are the same or similar [19]. In these cases, considering equity between DM opinions could lead to bias or distorted results. Consequently, some methods focus on deriving DM weight to assign differing importance at the opinion aggregation. These methods can be commonly used as multi-source information fusion techniques because the proposed algorithms can consider parallelism between DMs or informatic sources.

A huge number of studies to determine the weight of criteria can be found in the related literature, although there are limited studies determining the weight of experts [20,21]. Even so, some schemes have been proposed to divide and classify these methods. For example, they can be clustered in two groups: subjective and objective techniques. In subjective techniques, there is a supervisor or manager who directly assigns different weight to each expert, or the experts directly evaluate themselves. In objective techniques, mathematical approaches are used for this assignment. These approaches can use concepts such as distances between opinions or consistency between others. Another scheme proposed classification of objective methods [19]. This scheme divides methods by similarity of approach: index-based approaches, clustering-based approaches, and integrated approaches.

Nevertheless, these classifications are too broad and general to consider all current methods. In a scientific field where new approaches are constantly being proposed, and use of existing approaches extended by considering uncertainty and multiple data types, the lack of outline to locate the main operation of a particular method can overwhelm experts with information. Furthermore, lack of consensus on the nomenclature can lead to misreading or misunderstanding among inexperienced readers. It is common for multiple MCGDM techniques to use the same terminology, such as consensus index, support, similarity or credibility, to mean the same thing. At the same time, other academic papers may use the same terminology to refer to unrelated processes.

Additionally, an in-depth literature review has highlighted some points on MCGDM approaches that require attention. First, although social choice theory was one of the first methods to be developed, few references are made to it in the articles, even though they focus on the choice of alternatives. Second, the stage at which methods can be applied needs to be analysed, as there are clear differences between methods that do not require accurate information, techniques that need opinions and procedures that need to know the alternatives. Third, a study should explore whether these methods accept the existence of prior influence since it is rare that all decision-makers have equal importance.

This paper aims to provide a novel general GDM classification scheme and go into greater depth on the state-of-the-art concerning all experts' weight assignment methods. For this purpose, the weightassigned methods classification is further developed by a literature review. This new classification scheme will allow not only experts and scientists working in the field of MCGDM to know which methods exist but also explain the main features of each technique, such as the implementation phase, the data used and the main processes. Furthermore, multiple MCGDM critical points are analysed to highlight future research pathways, providing information on gaps in current methods in each proposed category.

The remainder of this paper is divided into the following sections. Section 2 presents the proposed GDM method classification. Section 3 presents the literature review methodology and main reviewed characteristics. Section 4 outlines the literature review. Section 5 reports the results analysis and discussion, explaining future research pathways and their challenges. Conclusions are drawn in the last section.

2. Classification of group decision methods

Although the study focus on DM weighting techniques, the literature review has made it possible to detect and classify multiple types of GDM practices. Consequently, a general GDM methods scheme is presented which classifies the decision makers' weighting techniques in detail. Future research can be applied to other clusters, although they are not part of this study. Other current schemes are presented in the first part of this section to help readers understand current classification practice.

2.1. Previous GDM classifications

Several classification schemes have been proposed to date. For example, one of the earliest divided methods into process- or contentoriented [22]. Process-oriented approaches are methods intended to generate new ideas to understand and structure the problem. Content-oriented approaches work with the opinions that are presented to aggregate them. This scheme was later improved with the second content-oriented layer [23] shown in Fig. 1, which divides the group into implicit multiple attribute evaluation, explicit multiple attribute evaluation and game-theory approaches.

Implicit multiple attribute evaluation refers to methodologies closely related to social theory techniques. Criteria, indicators, or requirements used for the selection are not explicitly stated by DMs. On the other hand, explicit multiple attribute evaluation clearly demonstrates the criteria and preferences of each DM. MCDM methods and their extensions fall into this category. Finally, game theory approaches are methods which follow traditional game theory. There are two game theory approaches: cooperative and non-cooperative. In GDM processes these methods mainly study coalitions among DMs.

Subsequently, the wide range of explicit multiple attribute evaluation methods made cluster upgrade mandatory. A new group named objective methods for deriving DM weight in GDM was created as part of explicit multiple attribute evaluation methods [19]. As shown in Fig. 2, this group is formed by similarity-based approaches, index-based approaches, clustering-based approaches, integrated approaches, and other approaches.

In this scheme, similarity-based approaches use the distance to weight the DMs. This distance can stem from DM assessments, or a fixed point usually named the ideal solution. Commonly, when a DM has a group-centered assessment it gains importance. Index-based approaches are divided by consensus or consistency approaches. The first group generates an index regarding the concept of "soft consensus" [24], which aims to improve the consensus between DMs allowing discussions or changes. The second group uses DMs' assessment consistency to weight them. The more consistent a DM, the more weight it receives.

Clustering-based approaches are commonly used with large groups of DMs. These methods usually divide the process into two layers. The first layer consists of an expert division by cluster, and each cluster receives an initial weight. The second layer assigns an individual expert weight inside the cluster. The importance of the final DM is derived from the aggregation of both weights. However, these layers use different weighting methods (i.e., cluster size for first layer and similarity between experts for the second layer) which came from other approaches. Finally, integrated approaches are generated by aggregating two or more previously presented methods, and other approaches refer to any



Fig. 2. Objective methods classification. Source: [19].



Fig. 3. Classification of GDM methods. Developed decision maker's weighting methods are shaded.

which do not belong in any category.

The aforementioned classification schemes present some limitations and inherent weaknesses. The scheme shown in Fig. 1 has an unduly broad approach regarding the large number of proposed methodologies. In GDM, techniques have been developed from multiple fields (i.e., psychology, economy, or politics), which demand a more accurate classification. Subjective or objective division has similar limitations. The scheme shown in Fig. 2 only considers objective methods and provides clusters that can be easily misread by inexperienced readers. As an example, soft consensus can be easily misread as similarity because one of the few differences is how weight evolves. Furthermore, cluster-based approaches are mainly generated by combining other methodologies, similar to integrated approaches.

2.2. Proposed classification of GDM methods

The proposed GDM methods classification scheme, shown in Fig. 3, is based on analysis of previous proposals and their limitations and shortcomings, found in the current GDM literature. This classification system encompasses all GDM methodologies in six groups, although only the DMs' weighting methods cluster (the core of this manuscript) is developed. This therefore provides a clear primary guideline for other categories, which will allow other authors to expand and complete the classification proposal with specialized studies.

Negotiation methods contain techniques based on sharing ideas with their corresponding discussion and debate, to reach an agreed solution. There is no need to state the selection criteria, or the process to choose alternatives. Clear examples are the DELPHI method and its extensions [25,26], devil's advocate [27] or the ward method [28]. Negotiation methods can be used to study how decision-making evolves in negotiation and propose various action guidelines.

The main aim of the weight assignment models is to define the importance of each DM. This category, which this manuscript aims to develop further, is divided into whether the analysis is carried out on the experts, on their evaluation or if these two methods are merged. In expert analysis methods, expert characteristics, interrelationships, capacities, or coalition power indexes are the basis for the weighting process:

- The first subdivision contains external assessor or interpersonal assessment (EAIA) techniques. In these processes, there is an external assessor (or manager) who assigns weight directly to an expert panel. Furthermore, some techniques propose an interpersonal assessment method in which the DMs weight each other. These practices are commonly known as subjective methods.
- The second subdivision is focused on the study of DMs' capabilities or the needs they present for the problem (CNA). These techniques generally use a common characteristic to compare the DMs such as experience and expertise, their university degree, or knowledge related to the problem field.
- The third and last subdivision is related to game theory approaches (GT). These methods are commonly used when a previous influence must be considered, and they study the DMs' ability to form coalitions or to be the main promoter of a decision.

On the other hand, assessment analysis methods only study the opinion, assessment, or position that each DM has taken in the problem. Although there are simple works based on the distance between the opinions and a known statistical value [10-12], the methodologies have been developed to consider criteria multiplicity. This development has led to a wide range of methodologies and consideration of the following typologies:

- Group position methods (GrP) weight DMs in terms of their assessment distances within the group. The main objective is to reduce the distance by modifying the opinion weight to obtain the closest position among them.
- Group centroid (GC) methodologies consider the existence of an "ideal solution", a real or virtual value which represents the group's desired assessment. This value is named the group centroid. DMs are weighted regarding the distance between their assessment and this centroid.
- Optimization techniques (O) are generally built on these first two methods. By programming algorithms or using feedback, an iterative process is carried out with the aim of minimizing distances. It is worth mentioning that many of these methods present the goal in a consensus index to be maximized, calculated as the inverse distance between points.
- Quality assessment (QA) techniques weight DMs according to the quality of their opinions, generally compared within the group. This quality is estimated by mathematic values of entropy, certainty, or consistency, among others.

Data-aggregation methods analyze evaluations given by the DMs to be able to add them directly using operators. These operators may be closely related to the GrP, QA and GC of the previous cluster; however, its procedures do not contemplate giving explicit weight to DMs. Therefore, in many cases there are the same type of studies in the calculation of entropy, certainty, and consistency, which could be considered QA if they openly relate this weight to the experts. The same happens in the distance to the mean operators, which are extremely similar to GC, without giving weight to the experts.

Although these operators can be as simple as direct aggregation or AHP-derived results [29], operators for uncertainty using multiple data



Fig. 4. Literature review gathering process.

types have also been developed, such as spherical fuzzy sets [30,31] or linguistic information [32–34]. These have become more important as they are included in the MCGDM field. Some authors use them as the main link to obtain input data for TOPSIS [35–37], ELECTRE [38,39], or TODIM [40] methods. Among others, there are operators that use the Dempster-Shafer theory of evidence (DST), in which each expert's evaluation is treated as evidence [41].

Ranking aggregation methods can be subdivided according to their theoretical basis. In a first group, there are set-ordering probability distribution techniques that follow the Thurstone scale and the law of comparative judgment [42] or binary comparisons [13]. Heuristic methods fall into a second group, which instead of seeking optimization, focuses on obtaining a solution by simple and intuitive mathematical approximations. For instance, this division includes Borda methods and their extensions [14], as well as studies of government elections [15] or Markov chains [43,16].

Finally, a third group of stochastic optimization methods can be generated. These techniques synthesize the scenarios given by all DMs based on distance optimization criteria. The ideal final ranking is generally obtained by minimizing the Kendall or Spearman distance between each iterative approach and the DMs' rankings. Some notable works in this field have been probabilistic approximations using Monte Carlo models [18], maximum probability Bayesian approximations [44] or uncertainty inclusion due to lack of reliability in incomplete rankings [45].

The final two groups are integrated and other methods. Integrated methods are generated by aggregating two or more previously presented methods, and other approaches refer to any that do not belong to any of the given categories.

It is important to emphasize the classification assigned for large group methods which use clustering systems. Commonly, these methods have been assigned to a specific group for them, as they use a doubleweighting layer. The first one usually assigns a weight regarding the cluster size. The second one uses one of the previously presented methods. Hence, in this GDM framework large group methods with clustering systems are classified by the second layer usage, or the most appropriate technique.

3. Literature review methodology

This section presents the literature review methodology. A systematic review involves a five-stage structure [46]: (i) formulation of the problem, (ii) determination of the data collection strategy, (iii) evaluation of the retrieved data, (iv) analysis and interpretation of the literature and finally, (v) presentation of the conclusions.

To address the paper's objective, the main question formulated in this study was (i): Which methods and techniques have been proposed to aggregate opinions and assessments in GDM environment? A data collection strategy was determined to answer this question. This strategy (ii) is shown in the following subsections, and it involves data gathering and a results classification process.

3.1. Gathering process

Articles were identified by the internationally-recognized bibliographic database Web of Science (WoS), which analyses articles from over 12,000 journals worldwide [47]. The use of this database was mainly justified by the depth of its coverage, yielding more outputs than any other database collection. Furthermore, it is a major multidisciplinary database of Thomson Reuters Web of Knowledge.

Having determined the database, two different research methods were used, as shown in Fig. 4. The first one focused on 5 general titles and topics related to GDM. The results obtained were restricted by a minimum of 1 citation ($c \ge 1$), given the general nature of the search. The second method centres on a specific search of closely associated DM weighting topics and a manual search in related journals. No minimum citation restriction was applied to the results.

13,717 papers were obtained thought the literature research method. These were later filtered to obtain the final set of articles on which to run the qualitative and quantitative analysis. The filtering process was conducted as follows: (i) discard duplicated articles; (ii) automatic eliminate papers if the title includes the words "operators" and "operator". Operator methods commonly use a direct aggregation without assigning weight to the DMs. (iii) automatic selection of papers if the title includes "consensus", "importance", "weights" or "experts". (iv) title and abstract analysis to detect potential results. (v) working frame analysis to identify DM weighting techniques.

As a result of this process, a final set of 208 references was selected for further analysis and interpretation. Among them, 201 different methodologies were identified and classified using the scheme mentioned above. The remaining 7 selected results concerning alreadyclassified methods.

It is highlighted that article categorization was only based on the DM weighting process which was presented. As multiple references contain

Data input information used in classified results.

Group	Acronym	Data input information
General	CI	Crisp Information
	ZI	Z-information
	II	Interval information
	InMI	Interval multiplicative information
	IMI	Intuitionistic multiplicative information
Fuzzy	FI	Fuzzy information
	GFSI	Generalized fuzzy soft information
	SFI	Spherical fuzzy information
	IFI	Intuitionistic Fuzzy information
	OFI	Ordered Fuzzy Information
	IIFI	Interval intuitionistic fuzzy information
	TFI	Triangular fuzzy information
	TIFI	Triangular intuitionistic fuzzy information
	TMI	Triangular multiplicative information
	TrFI	Trapezoidal fuzzy information
	TrIFI	Trapezoidal intuitionistic fuzzy information
	IITrFI	Interval intuitionistic trapezoidal fuzzy information
	TrCFI	Trapezoidal cubic fuzzy information
	HFI	Hesitant fuzzy information
	DHFI	Dual hesitant fuzzy information
	IHFI	Interval hesitant fuzzy information
	NWHFI	Normal wiggly hesitant fuzzy information
	P-HFI	Probabilistic hesitant fuzzy information
	PHFLI	Proportional hesitant fuzzy linguistic information
	DHHFLI	Double hierarchy hesitant fuzzy linguistic information
	PFI	Pythagorean fuzzy information
	PiFI	Picture fuzzy information
	q-ROFI	q-rung orthopair fuzzy information
	SNI	Single neutrosophic information
	INI	Interval neutrosophic information
	SLNI	Single linguistic neutrosophic information
	P-MNI	Probability multi neutrosophic information
Linguistic	LI	Linguistic information
	LHI	Linguistic hesitant information
	2TLI	2-tuple linguistic information
	I2TLI	Interval 2-tuple linguistic information
	IIULI	Interval intuitionistic uncertain linguistic information
	MGLI	Multi-granularity linguistic information
	P-LI	Probabilistic linguistic information

Table 2

Analysed convergence typologies.

Acronym	Convergence type
DSE	Expert decision
GP-DO	Distance optimization
GP-EDO	Entropy distance optimization
GP-IC	Consensus index optimization
GP-CsI	Consistency index optimization
GP-ECsI	Entropy and consistency index optimization

MCDM-applied methods, there are countless external variables such as criteria or attribute weight assignment that do not fall within the scope of the study.

3.2. Classification process

To help researchers and experts extend their knowledge regarding DMs' weighting methods, expanded cataloging tables are given in each proposed classification. These tables contain the reference name, year of publication and five main characteristics that have been identified as distinguishing methodology elements.

The first feature shows the input data typology. Since many authors use terminologies such as "values", "numbers" or "sets" to refer to the same source, the "information" ending was used to subdivide them. Table 1 contains the input data information with the related acronyms.

The second characteristic shows when the group decision process DMs receive their weight. Phase 0 methods only require knowledge of the experts and/or criteria involved in the process. These methods do not analyze the consensus solution as such, but the previous information. Phase 1 methods focus on DMs' opinions and assessments. Finally, phase 2 methods explicitly require knowledge of alternatives. These techniques use preference matrices, ordering of alternatives or final rankings for the analysis.

The third property is related to the variation in the DMs' assigned weight. Methods are classified regarding whether the weight is assigned statically (invariably) or dynamically (iteratively), changing throughout the process. If the weighting process is iterative, the fourth property determines which type of convergence marks the iteration ending. Generally, iterative processes are related to dynamic methods. However, in some individual cases, the process is iterative, but the weight assignment is static. Different types of convergence and their acronyms are shown in Table 2.

DSE convergences are given by an external expert or the group itself. In multiple analyses, the external expert is considered a moderator rather than a weight allocator for DMs. GP convergences refer to the common goal programming systems used by optimization methods. These weighting techniques are usually linked to a distance (deviation) minimization (GP-DO), a consensus index maximization (GP-IC) or a consistency index maximization (GP-CsI). In other cases, there are multiple system methods which use entropy minimization with consistency maximization (GP-ECsI), or entropy and distance minimization (GP-EDO).

It is important to emphasize that multiple authors use the same terminology for different purposes. Consensus indices are generally understood as distances between DMs' assessments, but also between them and the ideal solution. Therefore, methods that consider the consensus index as a distance between DMs will be contained in GP-DO. Methods that consider this index as the distance to the ideal solution will be presented as GP-IC.

Finally, the last characteristic is the ideal or objective solution consideration. Generally, this solution is considered the mean of the group, either arithmetic or geometric. However, some articles use general uncertainty approximations (RA), the fixed-point theory (FPT) or the Fermat-Torricelli point (FT) to detect optimal consensus points, or group centroids among others.

4. Results and classification

This section presents and classifies the retrieved data and corresponds to the third Cooper stage (1989). The following sub-sections coincide with the previously stated weighting methods for DMs from the proposed classification scheme.

Hence, selected papers are directly categorized according to their group. Furthermore, the aforementioned categorization tables are provided at the end of each sub-section. It is important to mention that the results are ordered by methodology similarities rather than publication year.

4.1. External assessor or interpersonal assessment (EAIA)

In external assessor assessment methods, Keeney and Kirkwood [48] propose the use of a weighted additive social welfare function with all DMs' opinions to aid the assessor with the weighting allocation. This idea was later developed by Kelemenis et al. [49] to allow multiple criteria. In that case, the assessor can give a different weight to each DM in each of the criteria.

Similar direct weighting approaches, in which the external assessor directly evaluates DMs, are proposed by Palomares and Martínez [50] and Tabatabei et al. [51], that name the external assessor as a manager or moderator. Hafezalkotob and Hafezalkotob [52] recommend that this manager assigns weight regarding DM talent, experience, and knowledge, like in a CNA methodology. Wang et al. [53] present a clustering method in which a second layer is assigned by the external assessor.

Other authors have developed tools to make the assessor's work

easier. For example, Wu et al. [54] develop a linear programming system to decide the DMs' weight based on a comparison. Other authors use the Analytic Hierarchy Process (AHP) method in uncertainty to assign weight in personnel selection [55,56], or propose that the project leader uses the best-worst method (BWM) among DMs [57]. Gupta et al. [58] propose a voting system for a panel of managers who weight a panel of DMs on intuitionistic fuzzy information. Yang et al. [59] transfer this voting process to a single external assessor worst-best pairwise comparison system. Finally, Sun and Huang [60] propose a gray correlation adjustment tool to be applied once the assessor has decided on the weight. Its main objective is to minimize the negative effects that these may have on each expert's objective.

Regarding interpersonal assessment methods, Bodily [61] presents a commitment model using Markov chains. This approach is based on how experts delegate power to external utility functions until a valid conclusion is reached for the whole team. However, experts' scoring methods are more common than delegation approaches.

Ramanathan and Ganesh [62] use the original AHP to allow DMs to assess each other based on problem contexts. Van der Honert [63] uses multiplicative AHP (MHP) and SMART techniques to develop REMEM-BRANT, an interpersonal evaluation software tool. Other models consider that score as a direct degree of confidence between experts [64, 65]. Liu et al. [66] use that degree of confidence assignation with a self-confidence index to allow a self-evaluation for each expert. However, this level of confidence among DMs can be analysed through social graph theories. Tian et al. [67] weight DMs regarding the centrality and connections of each DM's opinion, considering closeness as confidence.

Finally, some EAIA methods introduce a feedback system. Dong et al. [68] propose an interpersonal multi-criteria evaluation concerning professional skills, cooperation, or fairness, among others. If the level of consensus is not reached, feedback to DMs is required so that they can modify their opinions.

4.2. Capabilities and needs analysis (CNA)

The only study that considers assessments as needs is Brock's approach [69]. This author adopts a Nash theory approach to use Pareto's limits and frontiers to assign a fair weight to each DM.

On the other hand, many studies have used analysis of expert capabilities to assign importance within the group. Herowati et al. [70,71] propose a DM discrimination and inconsistency capability analysis through repetitive evaluations. Discrimination is also studied by Cheng et al. [72], who analyze the incomplete responses of DMs to assign them a degree of experience according to their preference matrices. Chakhar et al. [73] use the DMs' accuracy capability by comparing it against the team's pre-ordered groups of preference. Chunhua et al. [74] assign the weight considering the DMs' experience and knowledge regarding alternatives such as function, manufacturing, environment, economy, and society.

This experience has been widely applied in capability studies. Ivlev et al. [75] present a multi-attribute function considering each DM's years of experience and education. Liu et al. [76] continue along this line by scoring DMs based on their professional qualification and work experience. Borissova [77] uses a similar function considering experience and level of competence in the problem context. Elbarkouky and Fayek [78] send each DM a form to fill in that considers their years and diversity of work experience, position and time in the company, plus enthusiasm and willingness to participate. DMs are weighted by comparing the group forms. Sellak et al. [79] propose a degree of experience comprising the evaluation hesitation and DMs' interests and preferences presented in the preference matrices. Bai et al. [80] develop an online database text algorithm to define DMs' ability and experience based on their internet profiles, literature, and scientific achievements.

Other approaches are based on personal bias, self-confidence, and roles. For example, Chatterjee and Bhattacharyya [81] propose a probabilistic DM approach index based on three factors; capacity for

precision regarding the majority opinion, subjective biases shown and the difficulty of the problem to be solved. Slevin et al. [82] propose a form to state the self-confidence of each DM in their respective assessments. Finally, Liao et al. [83] develop a role study for a large group of experts. Once divided into clusters, if one DM has different characteristics from the majority, it obtains more weight to balance the decision from different points of view.

4.3. Game theory approaches (GT)

Game theory methods start from a previously-given influence to analyze the participant's power in a decision [84]. That power index (PI), understood as weight in this study, is considered the ability of a DM to be a decision promoter by coalition capability. GTs are usually divided into cooperative or non-cooperative or complete and incomplete depending on the exchange and type of information. However, the main characteristic in this study is how PIs are calculated. Hence, the main difference between methods is related to the consideration of winning coalitions.

Shapley and Shubik [85] show that all coalitions are possible, and PI is given by the number of times a DM generates a winning coalition, known as a pivot. Banzhaf [86] states that only winning coalitions should be used, and critical DMs should be studied. DMs are considered critical when a modification of their votes make the coalition lose its winning status. Deegan and Packel [87] introduce the minimum winning coalitions, which do not contain any unnecessary DMs. Felsenthal [88] modifies the minimum winning coalitions for minimum size winning coalitions as a winning DM aims to deal with the minimum number of possible participants in their coalition.

4.4. Group position methods (GrP)

Position among assessments can be considered a negative consensus effect. For example, some studies reduce DMs' weight if their evaluations are further from the others [89], calling it ineffectiveness by means of thinking errors [90]. On the contrary, Xu and Zhou [91] use the maximum deviation method to reduce DMs' weight. These authors consider that similar (closer) evaluations are ineffective to differentiate alternatives.

Nevertheless, assessment distances are generally considered a positive characteristic like compatibility [92] or correlation [93]. From these perspectives, the closest DMs receive the highest weight. To handle uncertain distances, Wu et al. [94] propose the use of a probability distribution with geodesic distance to approach the Manhattan distance. Lu et al. [95] incorporate the Dempster-Shafer theory (DST), in which assessments are considered evidence, into a probability function. Whang et al. [96] use the same theory in SLNI. Other authors [97,98] have used a similar DST procedure with Bayesian approaches [99]. This latest technique has been modified for use of TrCFI [100] or DHHFLI and q-ROFI [101,102].

Other GrP methods consider distance among assessments such as similarity [103–105], opinion transition facility among DMs [106,107], consensus degree index [108], or amount of support received [109,110]. Zang et al. [111] use the preference matrix distances to assign DMs' weight and introduce the results to ELECTRE. Liu and Cheng [112] use a similar P-MNI measurement to obtain the input values to MABAC. Chen and Zou [113] include a reassurance factor based on the similarity of evaluations between experts to reinforce their shared positions.

However, other approaches are proposed with viewpoints beyond negative or positive effects. Gupta et al. [114] consider the distance as an advantage or a disadvantage. The more address matches it has with other assessments, the higher advantage score an assessment will receive. French [115] studies the assessment modification in a discussion to assign influence to each DM, measured by the systematic position variations throughout a debate. This is translated into weight regarding the distance covered in each assessment. Subsequently, Pérez [116] expands this study to include social interaction graphs.

Distance variation throughout a debate has been also studied by other authors. Xue et al. [117] analyze DMs who have approached a third-party position to increase their weight, as their confidence has been reinforced. Fu et al. [118] propose that the experts' weight is given according to their reliability in a discussion process. In this approach, reliability is calculated according to the variation of the distances between assessments. Finally, Zhou et al. [119] use an approach which considers not only how DMs approach their position, but their persistence in defending their opinion.

4.5. Group centroid methods (GC)

Some authors have even studied the real centroid of a group assessment [12], where average is the most common option. Average deviation [120,121] and direct distance [122–124] are easy ways to consider a fair decision.

Nevertheless, average is also used as an indirect measurement. Ye [125] uses a scoring function to rate the differences between DM assessments and the average set, comparing them to assign the weight. Gitinavard et al. [126] use preference matrix deviations regarding the aggregate mean to obtain the least deviated matrix. This matrix is used to assign weight regarding original DMs matrixes.

Other authors appoint this distance as a 'similarity degree'. Chai et al. [127] use the similarity to assign the weight by comparison among DMs. Ye [128] uses a similar formulation to assign weight to DMs and criteria at the same time. This degree has been widely used to obtain TODIM input data regarding multiple information as TFI [129], P-HFI [130] or PLI [131] to adapt MCGDM to an existing MCDM.

These studies propose a double step; first, DMs receive an assigned weight and subsequently, the MCDM is applied. However, some studies have directly proposed a MCDM adaptation. Yue was inspired by Shih et al. [132] studies to introduce his first distance approaches [133], which led to his well-known extended TOPSIS or ETOPSIS [2]. Using a modified TOPSIS technique, this author develops a methodology based on closeness coefficients to adapt this MCDM to a MCGDM. Lately, this methodology has been adapted to II [134–136] and IFI [137], which have been used in other studies [138,139]. Other authors have extended Yue's work modifying input data to OFI [140], PFI [141], MGLI [142] or IFI with Hamming distance [143], proposing new closeness coefficients [144–146] or with the use of fuzzy set theory approaches instead of the arithmetic average [147] for the ideal solution.

Other TOPSIS-related works include a similar Yue's approach by Jiang y Wang [148], using IITrFI and the closeness coefficient, criteria disagreement between DMs [149], or HFI [150] and SCI adaptations [151].

Beyond existing MCDM adaptations, some novel MCGDM methods have been developed. Yue [152] presents a new method based on vectorial projections between experts and the ideal solution. Later, he extends this method to IFI [153]. As in ETOPSIS, other authors have worked with this method and have proposed new adjustments or extensions. Yang and Du [154] propose aggregating cosine for expert direction comparisons. Xu and Liu [155] and Sun [156] extend the method to II and IIFI respectively. Liao et al. adapt the method to IMI [157] considering the Euclidian and phycological distances, used by Luo et al. [158] to obtain MULTIMOORA input data.

Boix-Cots et al. [159] develop HIVES, a novel methodology which uses social theory constraints and statistical mathematics to assign DM weight. This technique uses the social ideal consensus point (SICP) as the ideal solution, which corresponds to the arithmetic mean of the second and third data quartiles. The value that this obtains ensures that bias is avoided and equals the optimal value for the majority. This method can manage previously assigned influence on each DM or source and maximizes the response (or DM satisfaction), as it analyses MCGDM problems criterion by criterion.

Finally, Gong et al. [160] propose to give the weight by obtaining

two factors based on the ranking analysis method, which studies the preference position of each expert with respect to their aggregate mean.

4.6. Optimization methods (O)

There are methods based on optimizing these GrP and GC distances. In these cases, the weight takes on a dynamic form until the distance has been minimized or a consensus index threshold is reached. Nevertheless, the consensus index is usually distance-based.

Among the GrP DMs distance or deviation optimization systems, some studies use programmed minimization models under the pretext of decreasing discordance and increasing consensus [161–165], or to increase the support index, generated by minimizing the distance among DMs [166]. On these, some authors have proposed approximation systems for the same purposes [167]. Other minimization models are focused on the quadratic distance between DMs' preference matrices [168,169].

There are techniques that focus analysis on the assessed alternatives. Meng et al. develop a quadratic programming model that maximizes the consensus index for each evaluated alternative [170]. Regarding rankings, Li et al. program a distance minimization system [171] and Zhang and Guo program a variance minimization system [172].

Some methods introduce feedback. For instance, Reagan et al. propose that DMs get a degree of respect according to the distance between them [173]. If the level of consensus with the final solution is not satisfactory, the opinion can be iteratively modified. Similar models are developed by Jin, Dong, and Cooper to minimize distances and improve consensus [174,175] through voluntary assessment modification.

Other optimization systems try to decrease the distance between DMs and the aggregated solution. In these cases, the consensus index uses this distance as a basis instead of the distances between DMs, as in GC. The clearest examples are programming dynamic weight in TOPSIS to minimize the distance to the ideal solution [176] in HFI and the minimization of distances to the aggregated solution through programming [177].

However, some studies consider that the aggregated matrix is calculated as the mean of the results obtained in each iteration, so they try to reduce their distances [178–180] or deviations [181]. Others consider the mean as the starting point [182] to calculate the initial weight, using iterative aggregation to calculate the optimization objective matrix [183,184]. Zhang and Xu consider that instead of the direct distance between matrices, the degrees of jointness and interaction should be used to perform the computation [185,186] according to the dominance of alternatives.

These differences between data are also known as discordance. Xu [187] develops two linear programming systems to minimize it by reducing the number of adjustments suffered by the DMs due to the weight, a technique that has been applied to obtain TOPSIS data [188]. TOPSIS has also been used in conjunction with programming models to minimize both distances; among DMs' assessments and between them and the aggregate solution [189].

Other authors use attribute values as a prior step. For example, Yu and Lai propose that DM weight is decided once the attributes are aggregated using an operator [190], to minimize the quadratic distance and discordance. One key point of this study is that the level of consensus is used to indicate whether the aggregation method should be varied. Chen et al. use this model as a basis for developing a minimum-maximum attribute weight optimization method [191]. Taking this operator use idea, Li et al. develop nonlinear programming in MATLAB for the minimization of discordances in a DM cluster [192] in an IVIF uncertainty environment. Among other characteristics, Wan et al. use the degrees of fuzzy membership for each assessment component to optimize distances by minimizing discordance [193].

Some studies propose that these methods should be accompanied by a feedback and re-evaluation system [194–196]. If the consensus indices do not exceed the thresholds, the DMs must modify their assessment at

the marked critical points, to iterate the process.

There are also techniques focused on large groups of DMs. Tan et al. propose that weight should be assigned according to consensus and consistency. Taking the mean as a reference, consensus is calculated as the difference between distances and consistency as the difference of the deviations in the linguistic evaluation [197]. If the minimums are not reached, feedback is sent to the most biased DMs. Rodriguez et al. and Tang et al. [198,199] propose that DM weight depends on both group size and the displayed cohesion, calculated analogously to the similarity with the cluster centroid. This centroid is initially calculated as a random preferred alternative. If a minimum consensus is not reached, individual and group feedback is generated so the opinions are modified.

Finally, there are methods with uncommon consensus- and rankingbased methods. In the former, Yang and He use the fixed-point theorem for the optimization calculation of DM weight [200]. According to this theorem, a solution will converge when the iterative change of DM weight is imperceptible. In the latter, Ben-Arieh and Chen present a consensus level based on the distance between the final aggregate ranking and DM rankings [201]. However, weight is modified according to its contribution to the group, measured as the level of consensus reached in the group without its participation.

4.7. Quality assessment methods (QA)

Quality assessment methods use comparison indices not based on distance, but on the properties of the assessments themselves to weight DMs. These indices can be optimized as Optimization methods are with distances. One of the most-frequently used is the consistency index developed by Saaty, used directly by several studies in this review in multiple data types such as CI [202], 2TLI [203], or IIFI [204].

The method proposed by Toloie-Eshlaghy and Farokhi is similar, based on the iterations needed to obtain the convergence vector [205]. Wu et al. [206] extend these studies by introducing an amplification parameter to the consistency of evaluations to differentiate experts with similar opinions. The optimization has also been introduced in consistency maximization. Xu and Cai [207] program an algorithm that maximizes group consistency by minimizing the weight of low-consistency evaluations. Liu et al. [208] present an optimization system adapted to the uncertainty arising from the alternatives themselves in autonomous bootstrapping systems. In this case, he uses an entropy and weight modification system capable of increasing DM consistency.

The Shannon entropy, understood as the lack of information contained in each of the evaluations, has also been a widely-used index [209–213] to assign weight to DMs. Cheng et al. [214] use a relative entropy minimization algorithm to deal with incomplete information, which makes it possible to find and minimize entropy distances. Çalı and Balaman [215] use the entropy to find the DMs' degree of divergence, subsequently using ELECTRE and VIKOR. Li et al. [216] use a scoring factor instead of the degree of divergence between assessments to relate entropy to the weight, used to introduce TODIM. This entropy has been used in conjunction with the hyper entropy, considering the result as DMs' personalities [217] in a LI, P-LI, and LHI study.

Multiple studies have proposed other quality assessment indices. For example, Jin et al. [218] use a Sugeno integral to analyze the information linked to evaluations to find its degree of certainty. Liao et al. [219] develop a scoring formulation to calculate the quality of fuzzy evaluation, named degree of hesitation in the expert's assessment. Wang et al. [220] interrelate the multiplicative preference matrices with the characteristic evaluation matrices using a quadratic logarithm to obtain a degree of information reliability. To solve the interrelated system, he presents a difference minimization algorithm that increases the degree of reliability.

Finally, there are cluster methods which use a double quality assessment. Ma et al. [221] use a combination of familiarity and cognitive reliability related to consistency. An optimization model to

find the minimum entropy of the group is used for the familiarity calculation. Then reliability is added, calculated as cluster consistency. Zou et al. [222] use entropy and consistency of the cluster members' evaluations, added to the cluster size, to obtain DM weight.

4.8. Integrated methods

The latter group corresponds to methods that incorporate two or more of the above-mentioned techniques. One of the most used combines DM distances and quality assessments. For instance, Qi et al. [223] use a quality assessment study with a degree of uncertainty, subsequently modified by a degree of divergence between evaluations to obtain the final DM weight. Liu et al. [224] use a similar idea to calculate large groups of DMs, which are weighted by a double-layer system by entropy and minimum variance method. Zhang and Chen [225,226] use the consistency and proximity of assessments. Wang et al. [227] use the similarity degree among DM rankings, modified by a degree of support for when these rankings differ in validity.

Some studies propose including optimization in these techniques. Chen et al. [228] use the consistency degree adding a consensus index as a benchmark in an iteration system. This consensus refers to distances between DMs, and if it is not less than the threshold, DMs must modify their evaluations. Zhang et al. [229] propose a method to maximize group evaluation consistency while minimizing DM distances. Qi et al. [230] propose that the weight is given by the quality of the evaluation, which is modified by a minimization algorithm for distances between DMs. There are similar studies which use these double programming methods to minimize the hamming distance between DMs and the aggregated mean and to minimize each DM's weighted entropy [231]. For high assessment uncertainty, Xu et al. [232] propose a double algorithm to minimize entropy and distances between DMs.

Quality assessment methods are also used with GC techniques. Pang et al. [233] use the uncertainty degree of assessments with their mean and limit matrix distances. Liu et al. [234] modify the attributes using the assessments' entropy, from which a media value is obtained to compare DMs. Liu et al. [235] use a double layer weight assignment for large groups of DMs. In this case, the first layer uses an entropy index, which is added to the minimum variance model index obtained in the second layer. Li et al. [236] develop two cross-entropy programming models which are simplified to exact equations to calculate experts' professional experience and expertise using the aggregate group mean as a reference. Even though it can be misread as a CNA method, professional experience is obtained by the entropy deviation and expertise by the direct distance among DM assessments and the media. A similar system is proposed by Pramanik et al. [237], with an ideal matrix composed of minimum and maximum values, regarding cost or benefit indicators, used to apply cross-entropy.

GrP are also used with GC techniques. Davoudabadi et al. [238] apply similarity among DMs to assign a first step weight, which leads to an objective matrix. Later, distance to the objective is applied to obtain the modified weight. Wan et al. [239] suggest adding the DM similarity degree technique to modify the ideal ETOPSIS solution.

A group of interesting integrations mix assessment analysis with expert analysis techniques. In this context, it is common to use GC. Chen et al. [240] aggregate a weight assigned by the problem organizers with a weight obtained by the distance with the geometric mean. Jabeur et al. [241] suggest assigning a weight regarding an interpersonal ranking of the importance, which is modified by a distance optimization system between individual and final rankings. Yang et al. [242] use an interpersonal AHP evaluation to obtain the aggregated matrix, to calculate the DM cosine degree of similarity and obtain VIKOR input data.

Other models use both mean matrices and minimum-maximum matrices. Liu et al. [243] propose to include initial subjective weight to modify ETOPSIS mean matrix. Mohagheghi's suggestion is similar, with an initial weight according to the importance in their area of expertise to modify TOPSIS values [244].



Fig. 5. Different terminologies used and their meaning.



Fig. 6. (A) Results published by decade and category. (B) Percentage dominance of each category by decade using cumulative results.

GrP appears in several studies using expert analysis techniques. Mianabadi and Afshar [245] integrate the evaluation of a group manager with the analysis of DM distances. Xu [246] extends Bodily's interpersonal evaluation contribution with a measure of linguistic preference deviations that modifies the initial weight according to similarity between DMs. Ölçer and Odabaşi [247] use a subjective comparison matrix which scores the importance of each expert, and the difference between their evaluations. Wan et al. develop a multi-objective programming to minimize distances between DMs and their thrust degree [248]. Bai et al. [249] propose the use of age, level of education, position, and experience to draw an initial weight, which he modifies according to DM distance.

There are proposed approaches which integrate both distance methods. Liu and Li [250] use an initial subjective weight, which is modified according to the distance between DMs. It is subsequently adjusted regarding the distance between the DM and the optimal Fermat-Torricelli center-point distance. Chen et al. [251] use the distance between DMs to obtain the modified Shapley index value, which is compared with the distance to their mean.

Finally, there are expert-expert analysis integration methods. Liu et al. [252] suggest adding an initial weight assigned by an organizer with a degree of self-confidence stated by the DMs themselves. Ren et al. [253] use the degree of professionalism to evaluate experts according to how they give information, adding assessment entropy methods.

5. Analysis and discussion

This section analyses the selected literature and corresponds to the fourth Cooper stage [46]. The analysis contains discussion on a generalized problem, a comparison and interpretation of the articles and their classifications, and an analysis of internal characteristics linked to

decision-making.

The pervasive problem that has been found throughout the literature has been the need to adopt a common framework in the methods of assigning weight to DMs. The lack of unanimity in their classification and in the existing categories promotes the existence of ambiguous studies. In this study, the author's point of view has been prioritized, so the explanations and terminologies used by the authors have been paramount for the classification. Also, the approaches presented in the studies have been determining for the category assignment.

For example, some EAIA methods encourage the external evaluator to give weight according to CNA capabilities. Another ambiguity is found in CNA methods that use QA bases but refer to the DM capacities. In this case, the discrimination and consistency analysis capabilities [70] or the contributions provided [73] could easily be interpreted as the second category.

The major sign of the currently general disagreement are these authors' terminologies. Beyond the framework, multiple uses and explanations of indices or characteristics that share the same root have been detected. The same data receives different names and explanations, as shown in Fig. 5, whether from subjective or objective sources.

Irrespective of whether the distance is calculated between direct assessments, using the preference matrix or by analysing the rankings, multiple terminologies refer to the same distance or even both. For novice researchers or non-experts in the field this can lead to misunderstandings and misconceptions.

Hence, a simplification beyond the 2.2 section classification should be adopted. Table 11 presents two indices that refer to the various distances being considered. The first is the consensus index among experts (CIE), which brings together all those terminologies that refer to the calculation among DMs. For example, one study can seek to improve the CIE by reducing the distances among experts or differences among



Fig. 7. Main MCDM methods appearing in the result's titles.



Fig. 8. Percentage distribution of ideal solutions.



Fig. 9. Percentage of results with stated social rules.

assessments' characteristics, such as entropy. Analogously, the second proposal is the consensus index to the solution (CIS). This index is used to unify the terminologies that refer to the calculation among each DM and the ideal solution, whether using distances or assessments' characteristics. Adopting these indices, or their inclusion in subsequent studies, might be of great help to the community.

Regarding the results comparison and interpretation, there has been a notable increase in MCGDM-proposed methods. In Fig. 6A, categorized results are shown by decade, from 1950 until after 2020. Fig. 6B uses that information to demonstrate the category dominance per decade.

It is interesting to note that the earliest dates back to 1950, when Kenneth Arrow's social choice theory (1951) seemed to spur interest in opinion aggregation methods. GT techniques quickly appeared to mainly solve economic power problems [85], and the first GrP aggregations [115] were proposed to solve discussion processes. This dominance was extended during the 1960s, when some authors discussed existing techniques and proposed new GT [86] and GrP [90] approaches.

In the 1970s, 1980s and 1990s, external [48] or interpersonal [61, 62] assessments, novel GT approaches [87], and CNA studies [69,82]



Fig. 10. Percentage of data used in each category.



Fig. 11. Percentage of articles applied in each phase for each category.

were proposed. They were all part of what we consider to be subjective methods.

However, as the 21st century began, a paradigm shift took place. The trend has shifted in favor of more objective methods for several reasons. Although expert analysis assessments may be easier to apply, EAIA methods can be subject to dishonesty or bias in interpersonal or supraevaluator assessments. In addition, unfairness can also arise in CNA and GT techniques when DMs have information from their competitors.

External assessor or interpersonal assessment methods cataloging table.

Author	Year	Data	Phase	Weight	Convergence	Consensus
Keeney and Kirkwood	1974	CI	Phase 0	Static	-	_
Kelemenis et al.	2011	CI	Phase 0	Static	-	-
Palomares et al.	2014	CI	Phase 0	Static	-	-
Tabatabaei et al.	2019	CI	Phase 0	Static	_	-
Hafezalkotob and Hafezalkotob	2017	TFI	Phase 0	Static	-	-
Wang et al.	2019	LI	Phase 0	Static	-	-
Wu Qun et al.	2019	2TLI	Phase 0	Static	-	-
Aly and Vrana	2008	TFI	Phase 0	Static	-	-
Samanlioglu et al.	2018	TFI	Phase 0	Static	-	-
Z.Chen et al.	2022	IITrFI	Phase 0	Static	-	-
Gupta et al.	2019	IFI/IIFI	Phase 0	Static	-	-
Yang et al.	2020	NWHFI	Phase 0	Static	-	-
Sun and Huang	2013	CI	Phase 1	Static	-	-
Bodily	1979	CI	Phase 1	Dynamic	DSE	-
Ramanathan and Ganesh	1994	CI	Phase 0	Static	-	-
Van der Honert	2001	CI	Phase 0	Static	-	-
Chen et al.	2018	LI	Phase 0	Static	-	-
Wu Jian et al.	2015	TFI	Phase 0	Static	-	-
Liu et al.	2019	2TLI	Phase 2	Static	-	-
Tian et al.	2019	I2TLI	Phase 0	Static	-	-
Dong et al.	2016	CI	Phase 0	Dynamic	GP-IC	_

Table 4

Capabilities and needs analysis methods cataloging table.

Author	Year	Data	Phase	Weight	Convergence	Consensus
Brock	1980	CI	Phase 1	Static	-	-
Herowati et al.	2014	CI	Phase 2	Dynamic	DSE	-
Cheng et al.	2018	LI	Phase 2	Static	-	-
Chakhar et al.	2016	CI	Phase 2	Static	_	-
Chunhua et al.	2020	IFI	Phase 0	Static	-	-
Ivlev et al.	2015	CI	Phase 0	Static	-	-
Liu et al.	2019	LI	Phase 0	Static	-	-
Borissova	2018	CI	Phase 0	Static	-	-
Elbarkouky and Fayek	2011	FI	Phase 0	Static	_	-
Sellak et al.	2019	HFI	Phase 2	Static	-	-
Bai et al.	2017	CI	Phase 0	Static	-	-
Chatterjee and Bhattacharyya	2017	CI	Phase 2	Static	-	Majority
Slevin et al.	1998	LI	Phase 1	Static	-	-
Liao et al.	2020	CI	Phase 2	Static	-	Majority

Table 5

Game Theory methods cataloging table.

5		U	0			
Author	Year	Data	Phase	Weight	Convergence	Consensus
Shapley and Shubik	1954	CI	Phase 0	Static	-	-
Banzhaf	1965	CI	Phase 0	Static	-	-
Deegan and Packel	1978	CI	Phase 0	Static	-	-
Felsenthal	2016	CI	Phase 0	Static	-	-

Presenting unrealistic needs, use of certain criteria that favor some DMs in advance or the preconception of coalitions through the delegation of power can lead to a biased consensus.

Therefore, assessment analysis methods which ignore DMs and only look at their evaluation have been proposed as possible solutions to these problems. These techniques have grown exponentially since 2000, with major contributions on GrP, O, QA, and GC procedures, and the latter has been particularly noteworthy.

In fact, this development of assessment analysis methods might be directly related to the growing interest shown in MCDM methods. As the use of MCDM in multiple alternative analysis has recently expanded due to its objectivity, adding these methods to group decisions gives the process further objectivity. Several articles use their proposed methods in conjunction with MCDM. Fig. 7 shows the MCDM used regarding each category. Even though numerous MCDMs have been applied TOPSIS and VIKOR are the most frequently used techniques, while other methods have few appearances. This fact is surprising, considering the large number of MCDM techniques constantly applied in all scientific fields. Regarding categories, the GC wide application is remarkable. This is mainly due to Yue's work on Extended TOPSIS and Projection, which appear multiple times in the categorization. On the other hand, the lack of GC techniques applied with VIKOR is remarkable.

It is precisely this widespread GC use that leads the discussion to another interesting point. Although multiple possibilities have been proposed as an ideal solution, data shown in Fig. 8 make the general preference clear. This figure considers methods which use an ideal solution and separates the percentages according to which point is considered.

The mean is by far the most used statistical point to assume consensus by 56 out of 67 articles, representing an astonishing 83.58%. On the other hand, few articles use other approximations and percentages are widely distributed. Two articles (2.99%) use the mean affected by a previous weight [243,244] and 4 (5.97%) consider the consensus by majority [81,83,235,246]. RA [147], FTP [200], SICP [159], Max/Min [237], and FT [250] are only used by 1 article each, representing 1.49% per approximation.

The high degree of acceptance of the mean as an ideal solution is striking, even though some works oppose its unquestionable use [12]

Group Position methods cataloging table.

Author	Year	Data	Phase	Weight	Convergence	Consensus
S. Wan et al.	2021	TMI	Phase 2	Static	-	-
Theil	1963	CI	Phase 1	Static	-	-
Xu and Zhou	2017	P-HFI	Phase 2	Static	-	-
Zeng et al.	2016	IFI	Phase 2	Static	-	-
Thong et al.	2020	INI	Phase 1	Static	-	-
Wu et al.	2019	HFI	Phase 2	Static	-	-
Lu et al.	2007	CI	Phase 2	Static	-	-
Wang et al.	2019	SLNI	Phase 2	Static	-	-
Gupta et al.	2016	TrIFI	Phase 2	Static	-	-
Wan et al.	2013	TIFI	Phase 2	Static	-	-
Fahmi et al.	2019	TrCFI	Phase 2	Static	-	-
Krishankumar et al.	2020	DHHFLI	Phase 2	Static	-	-
Krishankumar et al.	2020	q-ROFI	Phase 2	Static	-	-
Meng et al.	2020	IFI	Phase 2	Static	-	-
Wang et al.	2019	ZI	Phase 2	Static	-	-
Zhang and Wang	2017	MGLI	Phase 2	Static	-	-
Wan et al.	2013	IIFI	Phase 2	Static	_	-
Xu et al.	2019	FI	Phase 2	Static	_	-
S. Wan et al.	2022	LI	Phase 2	Static	_	-
Liu et al.	2019	q-ROFI	Phase 2	Static	_	-
S. Wan et al.	2021	P-LI	Phase 2	Static	_	-
Zhang et al.	2020	PiFI	Phase 2	Static	_	-
Liu and Cheng	2020	P-MNI	Phase 2	Static	-	-
Chen and Zou	2020	GFSI	Phase 2	Static	-	-
Zhang et al.	2018	2TLI	Phase 2	Static	_	-
Xu et al.	2019	CI	Phase 2	Static	-	-
Xu et al.	2020	CI	Phase 1	Static	_	-
Gupta et al.	2018	IIFI	Phase 1	Static	_	-
French	1956	CI	Phase 1	Static	DSE	-
Pérez et al.	2016	CI	Phase 2	Static	DSE	-
Xue et al.	2020	CI	Phase 2	Static	-	-
Fu et al.	2015	II	Phase 2	Static	_	-
Zhou et al.	2020	CI	Phase 2	Static	-	-

and propose alternatives that improve consensus or present the potential loss of information produced by its use [163]. It should be noted that many methods do not use the mean as a direct solution, but as an approximation to the solution. This may be due to the implicit acceptance of social choice theory axioms. These axioms are used to force mathematical methods to contain a social basis that gives them some sociological weight and avoids situations of injustice, promoting equality.

However, even though scarce studies present social considerations in the procedures as shown in Fig. 9, many lack their explicit incorporation, making it difficult to find out about their contributions or to monitor them. Only 12 out of 201 categorized results (5.97%) clearly show a social rule, even though these methods are developed to help DMs with their alternative selection decisions. When focussing on expert analysis methods, 8 out of 39 (20.51%) articles show this feature. GT methods present 4 out of 4 (100%) for this characteristic, because these approaches study the interactions among DMs and were developed with a strong social basis. Regarding other categories, EAIA presents the characteristic in 3 out of 21 (14.29%) articles and CNA in 1 out of 14 (7.14%). It is remarkable that EAIA [61,48,62] and CNA [69] results are all from the last century.

Nevertheless, the assessment analysis results are astonishing. Only 4 out of 132 (3.03%) articles present social considerations, even these methods have been proposed to improve the objectivity and fairness of GDM, which are key points in the social choice theory approaches. Only 3 out of 33 (9.09%) GrP and 1 out of 38 (2.63%) GC articles have this characteristic. Like the previous case, 2 of the GrP results [115,90] are from the last century, while the latter [112] and the GC result [159] are relatively novel approaches. Finally, there are no integrated methods with this feature.

Instead, the articles contain extensive and detailed introductions to the type of data being used. These introductions are explained by the great MCDM data diversity. The increased complexity of MCDMs is linked to the further development of input data that allow DMs to express themselves in different ways. Moving away from exact positive values, the studies present possibilities such as ranges, uncertainty, or linguistics, as discussed in the characteristic's tables. Fig. 10 is presented using Section 4 tables data. This diagram shows the percentage of data typology according to each of the proposed classifications.

A clear, exact value dominance can be observed in most categories, except for GC and Integrated methods. This may be because the main objective of this study has been to classify new methods, which are usually presented with exact values. Subsequently, their complexity increases with new data. As many GC methods use the aggregate mean, several proposals are based on modifying the data used on this value. One clear example is the multiple input data modifications of methods such as ETOPSIS and projection. Integrated methods, on the other hand, generally start from combining existing methods and can therefore move on to other data. It is remarkable that all the GT methods are concentrated in the CI category. The scarce information obtained by the search process should be mentioned, and this would have to be expanded.

The last comparison and interpretation of results and their classification tables is the point at which these methods are applied in the GDM process. Fig. 11 shows the articles phase application percentages regarding each category. It is noted that phase 0 articles are contained in the expert analysis methods. As the classification proposal indicates, these techniques do not need to know either the alternatives or the assessors' opinions on the criteria or attributes. Therefore, it is normal that this phase does not appear with assessment analysis methods. Perhaps the most striking value is the whole phase 0 in GT techniques, as these are focused on analysing previous DM weight (Tables 3–10).

Nevertheless, assessment analysis methods also raise an interesting discussion point. Even though they are developed to work with phase 1 and 2 to improve the process objectivity, phase 2 dominance is remarkable. There are many directly related articles on multiplicative preference matrices or similar (116 out of 132), and few focus solely on

GC methods cataloging table.

Author	Year	Data	Phase	Weight	Convergence	Consensus
Tong and Wang	2016	IFI	Phase 2	Static	_	Mean
Zhang et al.	2016	2TLI	Phase 2	Static	_	Mean
Chen et al.	2021	HFI	Phase 2	Static	-	Mean
Li y Chen	2014	TrCFI	Phase 2	Static	-	Mean
Lin et al.	2018	HFI	Phase 2	Static	-	Mean
Ye	2013	IIFI	Phase 2	Static	-	Mean
Gitinavard et al.	2018	IHFI	Phase 2	Static	-	Mean
Chai et al.	2013	IIFI	Phase 2	Static	-	Mean
Ye	2014	SNI	Phase 2	Static	-	Mean
Qin et al.	2017	TIFI	Phase 2	Static	_	Mean
Zhang et al.	2019	P-HFI	Phase 2	Static	_	Mean
Wei y Wu	2019	P-LI	Phase 2	Static	_	Mean
Yue Z.	2011	IIFI	Phase 2	Static	_	Mean
Yue Z.	2011	CI	Phase 2	Static	_	Mean
Yue Z.	2011	II	Phase 2	Static	-	Mean
Yue Z.	2012	II	Phase 2	Static	-	Mean
Yue Z.	2012	II	Phase 2	Static	_	Mean
Yue Z.	2014	IFI	Phase 2	Static	_	Mean
Kacprzak	2019	OFI	Phase 2	Static	_	Mean
Biswas and Sarkar	2019	PFI	Phase 2	Static	_	Mean
Liu et al.	2013	MGLI	Phase 2	Static	_	Mean
Wan et al.	2015	IFI	Phase 2	Static	_	Mean
Dorfeshan and Mousavi	2019	PFI	Phase 2	Static	_	Mean
Gitinavard et al.	2016	HFI	Phase 2	Static	-	Mean
Saffarzadeh et al.	2020	II	Phase 2	Static	-	Mean
Yang et al.	2017	CI	Phase 2	Static	-	RA
Jiang and Wang	2014	IITrFI	Phase 2	Static	-	Mean
Hamdani	2017	CI	Phase 2	Static	-	Mean
S. Wan et al.	2020	HFI	Phase 2	Static	-	Mean
E. Aydoğdu et al.	2023	SFI	Phase 2	Static	-	Mean
Yue Z.	2012	CI	Phase 2	Static	-	Mean
Yue Z.	2013	IFI	Phase 2	Static	-	Mean
Yang and Du	2015	CI	Phase 2	Static	-	Mean
Xu and Liu	2013	II	Phase 2	Static	-	Mean
Sun	2015	IIFI	Phase 2	Static	-	Mean
Liao et al.	2018	IMI	Phase 2	Static	-	Mean
Boix-Cots et al.	2023	CI	Phase 1	Static	-	SICP
Gong et al.	2019	CI	Phase 2	Static	-	Mean

opinion analysis (16 out of 132). Furthermore, 6 out of these 16 articles are related to QA, as the method compares assessment characteristics. Thus, only 5, 1 and 4 phase-1 articles are GrP, GC and O, respectively.

There are many cases where alternatives are not known or may vary. Therefore, methods that might be applied without awareness of the alternatives (phase 0 and 1) are an interesting aspect to develop. Remember that methods using alternatives have been given phase 2 classification. Many of them could easily be adapted to lack of knowledge.

Finally, two internal method characteristics have been analysed due to their close connection with real-world GDM. Previous influence acceptance is one. In actual decision-making, it is not unusual for experts to differ in importance. Examples of this might be different business roles, or differences between DMs' experience or knowledge. The other is a component division process. Some studies have stated that dividing a multiple-criteria problem into criterion-by-criterion analysis could improve the consensus group response [3]. Hence, Fig. 12 shows the percentage of articles which use or comment on each of these characteristics.

The previous influence percentages are striking. Although it is a key point in GDM, it has not been widely considered, with a 4.48% (9 out of 201 results) occurrence. If GT data is ignored, as these methods are specialized in previously assigned influence, the value falls to 2.54% (5 out of 197 results) occurrence. While expert analysis methods could manage prior influence more easily (e.g., through external advisor consideration or role contemplation), assessment analysis technique methods require a structured system to consider it. This explains why these low percentage values are so remarkable. Even so, there are articles which consider it. Liu et al. [109] use the GrP approach as corrector

coefficient to weight given previously. In GC methods, Hamdani and Aydoğdu [151,149] only mention its existence while Boix-Cots et al. [159] develop a system that allows HIVES with the capability for previous influence treatment. Finally, in Integrated methods, the Shapley power index is used as the previous influence [251].

Regarding component division, the same number of articles demonstrate this characteristic (9 out of 201 results). The most common procedure is to assign a different weight to each DM regarding each assessed criterion or attribute in EAIA [49], GC [127,130], O [177], and Integrated [233,239] methods. Other systems have also been proposed. For example, in GC there is a system which modifies previous DM-assigned weight regarding how many of them have chosen to use certain criteria, allocating a DM weight-per-criteria [149]. In the same category, a technique has been proposed which analyses each component separately and subsequently aggregates the results [159]. Finally, a novel approach that assigns membership weight to each DM respecting each criterion [253] has been presented in the integrated category.

6. Conclusion

In this paper, a comprehensive analysis of GDM methods has been carried out to provide an in-depth state-of-the-art understanding on decision makers' weight assignment methods. In addition, the knowledge gained through the literature review has made it possible to propose a new classification scheme for all existing and future MCGDM methods. The development of this classification will allow scientists developing new methods to place their approach within a classification group, thus gaining insight into existing advances in the field. Furthermore, for the first time, experts will be able to get a clear picture of the

Optimization methods cataloging table.

Author	Year	Data	Phase	Weight	Convergence	Consensus
Li et al.	2017	HFI	Phase 2	Dynamic	GP-DO	-
Lin and Wang	2018	IFI	Phase 2	Dynamic	GP-DO	_
Ma et al.	2020	MGLI	Phase 2	Dynamic	GP-DO	-
Xu and Wu	2013	2TLI	Phase 2	Dynamic	GP-IC	-
Xu and Cai	2012	CI	Phase 2	Dynamic	GP-DO	-
S. Wan et al.	2021	InMI	Phase 2	Dynamic	GP-DO	-
Abootalebi et al.	2018	CI	Phase 2	Dynamic	GP-DO	-
Xu et al.	2016	LI	Phase 2	Dynamic	GP-DO	-
Xu et al.	2017	HFI	Phase 2	Dynamic	GP-DO	-
Meng et al.	2017	IIFI	Phase 2	Dynamic	GP-DO	-
Li et al.	2015	CI	Phase 2	Dynamic	GP-DO	-
Zhang and Guo	2016	CI	Phase 2	Dynamic	GP-DO	-
Regan et al.	2006	CI	Phase 2	Dynamic	GP-IC	-
Dong y Cooper	2016	CI	Phase 2	Dynamic	GP-DO	-
Ji et al.	2021	CI/FI	Phase 2	Dynamic	GP-DO	-
WC. Zou et al	2022	HFI	Phase 2	Dynamic	GP-DO	Mean
A. H. Liu et al.	2021	LI	Phase 2	Dynamic	GP-DO	Mean
Meng et al.	2016	I2TLI	Phase 2	Dynamic	GP-DO	Mean
S. Wan et al.	2022	P-LI	Phase 2	Dynamic	GP-DO	-
Wan et al.	2016	IIFI	Phase 2	Dynamic	GP-IC	Mean
Zhang et al.	2020	q-ROFI	Phase 1	Dynamic	GP-DO	Mean
Lee	2002	FI	Phase 1	Dynamic	GP-DO	Mean
Li and Wang	2019	HFI	Phase 1	Dynamic	GP-DO	-
Parreiras et al.	2010	LI	Phase 1	Dynamic	GP-DO	-
Zhang and Xu	2014	IFI	Phase 2	Dynamic	GP-IC	Mean
Zhang and Xu	2015	IIFI	Phase 2	Dynamic	GP-IC	Mean
Xu	2018	IIFI	Phase 2	Dynamic	GP-IC	Mean
S. Wan et al.	2021	P-LI	Phase 2	Dynamic	GP-DO	-
Yu and Lai	2011	CI	Phase 2	Dynamic	GP-IC	Mean
Chen et al.	2015	CI	Phase 2	Dynamic	GP-IC	-
Li et al.	2017	IIFI	Phase 2	Dynamic	GP-DO	Mean
Wan et al.	2017	IIFI	Phase 2	Dynamic	GP-IC	-
Dong et al.	2016	CI	Phase 2	Dynamic	GP-IC	-
Pang et al.	2017	IIFI	Phase 2	Dynamic	GP-IC	Mean
Zhang et al.	2016	TFI	Phase 2	Dynamic	GP-IC	-
Tan et al.	2020	CI	Phase 2	Dynamic	GP-IC	Mean
Rodríguez et al.	2018	HFI	Phase 2	Dynamic	GP-IC	-
Tang et al.	2020	CI	Phase 2	Dynamic	GP-IC	-
Yang and He	2021	IIFI	Phase 2	Dynamic	GP-DO	FPT
Ben-Arieh and Chen	2006	IIFI	Phase 2	Dynamic	GP-IC	Mean

Table 9

Quality Assessment methods cataloging table.

Author	Year	Data	Phase	Weight	Convergence	Consensus
Wu et al.	2018	CI	Phase 1	Static	-	-
Cabrerizo et al.	2010	2TLI	Phase 2	Static	-	-
S. Wan et al.	2020	IIFI	Phase 2	Static	-	-
Toloie-Eshlaghy and Farokhi	2011	CI	Phase 1	Static	-	-
Wu et al.	2012	CI	Phase 1	Static	-	-
Xu and Cai	2014	II	Phase 2	Dynamic	GP-CsI	-
Liu et al.	2013	IFI	Phase 1	Dynamic	GP-CsI	-
Hashemi et al.	2018	IFI	Phase 1	Static	-	-
RazaviToosi and Samani	2019	TFI	Phase 2	Static	-	-
Wu et al.	2020	LI	Phase 2	Static	-	-
Ye Jun	2013	IIFI	Phase 2	Static	-	-
Yue Chuan	2017	CI	Phase 2	Static	-	-
Cheng et al.	2017	LI	Phase 2	Dynamic	GP-EDO	-
Cali and Balaman	2019	IFI	Phase 2	Static	-	-
Li et al.	2015	IIFI	Phase 2	Static	-	-
Mao et al.	2022	LI / P-LI / LHI	Phase 2	Static	-	-
Jin et al.	2018	FI	Phase 1	Static	-	-
Liao et al.	2019	HFI	Phase 2	Static	-	-
Wang et al.	2013	II	Phase 2	Dynamic	GP-CsI	-
Ma et al.	2019	LI	Phase 2	Dynamic	GP-ECsI	-
Zhou et al.	2011	CI	Phase 2	Dynamic	GP-ECsI	_

state of play of MCGDM methods, clearly displaying the main features of their procedures and the data they work with. Furthermore, six issues for future research pathways have been highlighted:

Furthermore, six issues for have been presented to help the research community, novel researchers, or non-experts in GDM. Second, the use of the mere mean as direct ideal solution should be

First, the need for a unified working framework has been outlined. Ambiguity, misunderstandings, or simple misreading can arise if

Second, the use of the mere mean as direct ideal solution should be avoided. Several authors have questioned the veracity of this objective

multiple, interlocking concepts continue to be used. Hence, two indices

Integrated methods cataloging table. The methods characteristic shows the techniques used for each item.

Author	Year	Data	Phase	Weight	Convergence	Consensus	Methods
Qi et al.	2015	IIFI	Phase 2	Static	-	-	GrP + QA
Liu et al.	2015	2TLI/I2TLI	Phase 2	Static	-	-	GrP + QA
Zhang and Chen	2019	HFI	Phase 2	Static	-	-	GrP + QA
Wang et al.	2015	CI	Phase 2	Static	-	-	GrP + QA
Chen et al.	2015	IIFI	Phase 2	Dynamic	GP-DO	-	O + QA
Zhang et al.	2020	LI	Phase 2	Dynamic	GP-DO	-	O + QA
Qi et al.	2018	IHFI	Phase 2	Dynamic	GP-DO	-	O + QA
Z. Chen et al.	2023	IIFI	Phase 2	Dynamic	GP-DO + ECsI	-	O + QA
Xu et al.	2020	P-LI	Phase 2	Dynamic	GP-EDO	-	O + GrP
Pang et al.	2020	IIULI	Phase 2	Static	-	Mean	GC + QA
Liu et al.	2013	IFI	Phase 2	Static	-	Mean	GC + QA
Liu et al.	2017	IIFI	Phase 2	Static	-	Majority	GC + QA
Li et al.	2020	IIFI	Phase 2	Static	-	Mean	GC + QA
Pramanik et al.	2018	SNI	Phase 2	Static	-	Max/Min	GC + QA
Davoudabadi et al.	2020	IIFI	Phase 2	Static	-	-	GrP + GC
Wan et al.	2015	IIFI	Phase 2	Static	-	Mean	GrP + GC
Chen et al.	2018	CI	Phase 1	Static	-	Mean	EAIA + GC
Jabeur et al.	2004	CI	Phase 2	Dynamic	GP-IC	-	EAIA + O
Yang et al.	2019	PHFLI	Phase 1	Static	-	-	EAIA + GC
Liu et al.	2016	CI	Phase 2	Static	-	Mean*	EAIA + GC
Mohagheghi et al.	2017	IIFI	Phase 2	Static	-	Mean*	CNA + GC
Mianabadi and Afshar	2008	FI	Phase 1	Static	-	-	EAIA + GrP
Xu	2008	LI	Phase 2	Static	-	Majority	EAIA + GrP
Ölçer and Odabaşi	2005	TrFI	Phase 1	Static	-	-	EAIA + GrP
S. Wan et al.	2022	LI	Phase 2	Dynamic	GP-DO + IC	-	EAIA + O
Bai et al.	2020	TrFI	Phase 1	Static	-	-	CNA + GrP
Liu and Li	2015	II	Phase 2	Static	-	FT	EAIA + GRP + GC
Chen et al.	2021	q-ROFI	Phase 2	Static	-	Mean	GrP + GT + GC
Liu et al.	2019	LI	Phase 2	Static	-	-	CNA + EAIA
Ren et al.	2018	DHFI	Phase 2	Static	-	-	CNA + QA

*The mean has been modified by previously assigned weight.

Table 11

mucz prope		
Acronym	Index	Meaning
CIE	Consensus Index among Experts	Distance among experts
CIS	Consensus Index to Solution	Distance among the ideal solution and experts



Fig. 12. Percentage of articles which state previous influence or component division by category.

statistic and have analysed other possibilities. Others have applied novel procedures or proposed the use of other, more suitable geometric points.

Third, it was found to be necessary to consider concepts and constraints from sociological theories, such as social choice theory. Since this is a process of selecting alternatives or possibilities from among a group of experts, it seems obvious that there should be social guidelines for equal opportunities. However, most methods do not consider them.

Fourth, novel methods must also promote approaches in the absence of alternatives knowledge. In a world where possibilities are growing and alternatives might have multiple modifications, it is not realistic to consider that the alternatives will be known. Fifth, there is a need to consider including initial influence and separation of components. It is precisely the increasing complexity of today's world problems which leads to considering multiple experts or data sources and increasingly more criteria. These experts or sources may not have the same weight, and some studies show an improvement in the decision when analysing the criteria separately.

Finally, MCDM techniques and data types being used in GDM methods have been analysed to identify where future research might be focused. Providing a wide range of possibilities for experts to decide which method is more suitable will improve the use rate for these techniques and can open up new avenues of research by detecting which methods need to be adapted to certain types of data.

Nevertheless, the limitations of the present study should be mentioned. Firstly, although more than 13,500 papers have been analysed, many others may have yet to be included in the study. For example, contributions in the field are diluted within much more extensive studies, which treat this aggregation as secondary and focus on other results, such as the criteria and rankings obtained. Another limitation is the length of the manuscript, which makes it challenging to increase the number of comparisons between algorithms.

In future research, other groups of the proposed classification scheme, such as trading techniques or data aggregators, should be developed, expanding and complementing it. Additionally, future studies should analyze the algorithms used by each technique, comparing them with other approaches in the same classification group.

CRediT authorship contribution statement

David Boix-Cots: Conceptualization, Data curation, Formal analysis, Writing – original draft. **Francesc Pardo-Bosch:** Resources, Supervision, Validation. **Pablo Pujadas:** Resources, Supervision, Validation.

Declaration of Competing Interest

The authors declare that they have no known competing financial

interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Acknowledgments

The first author acknowledges support from the Spanish Ministry of Universities [grant number FPU18/01471]. The second and third author wish to recognize their support from the Serra Hunter program. Finally, this work was supported by the Catalan agency AGAUR through its research group support program (2017SGR00227). This research is part of the R&D project IAQ4EDU, reference no. PID2020-117366RB-I00, funded by MCIN/AEI/10.13039/ 501100011033.

References

- D. Boix-Cots, F. Pardo-Bosch, A. Blanco, A. Aguado, P. Pujadas, A systematic review on MIVES: a sustainability-oriented multi-criteria decision-making method, Build. Environ. 223 (2022), 109515, https://doi.org/10.1016/j. buildenv.2022.109515.
- [2] Z. Yue, A method for group decision-making based on determining weights of decision makers using TOPSIS, Appl. Math. Model. 35 (4) (2011) 1926–1936, https://doi.org/10.1016/j.apm.2010.11.001.
- [3] N.L. Kerr, R.S. Tindale, Group performance and decision making, Annu. Rev. Psychol. 55 (2004) 623–655, https://doi.org/10.1146/annurev. psych 55 090902 142009
- [4] I. Lorge, H. Solomon, Two models of group behavior in the solution of eurekatype problems, Psychometrika 20 (2) (1955) 139–148, https://doi.org/10.1007/ BF02288986.
- [5] D. Black, Theory Committees and Elections, Cambridge Univ. Press, 1958.
- [6] K.J. Arrow, Social Choice and Individual Values, Yale Univ. Press, 1963.
 [7] I.D. Steiner, Models for inferring relationships between group size and potential group productivity, Behav. Sci. 11 (4) (1966) 273–283, https://doi.org/10.1002/bs3830110404
- [8] I.D. Steiner, Group Process and Productivity, Academic Press, Ed, 1972.
- [9] J. Davis, Group decision and social interaction: a theory of social decision schemes, Psychol. Rev. 80 (2) (1973) 97–125.
- [10] H.W. Crott, K. Szilvas, J.A. Zuber, Group decision, choice shift, and polarization in consulting, political, and local political scenarios: an experimental investigation and theoretical analysis, Organ. Behav. Hum. Decis. Process 49 (1) (1991) 22–41, https://doi.org/10.1016/0749-5978(91)90040-Z.
- [11] J.H. Davis, L. Hulbert, W.T. Au, X.P. Chen, P. Zarnoth, Effects of group size and procedural influence on consensual judgments of quantity: the example of damage awards and mock civil juries, J. Pers. Soc. Psychol. 73 (4) (1997) 703–718, https://doi.org/10.1037/0022-3514.73.4.703.
- [12] V.B. Hinsz, Group decision making with responses of a quantitative nature: the theory of social decision schemes for quantities, Organ. Behav. Hum. Decis. Process. 80 (1) (1999) 28–49, https://doi.org/10.1006/obhd.1999.2853.
- [13] L.L. Thurstone, The method of paired comparisons for social values, J. Abnorm. Soc. Psychol. 21 (4) (1927) 384–400, https://doi.org/10.1037/h0065439.
- [14] J.A. Aslam, M. Montague, Models for Metasearch, SIGIR Forum (2001) 276–284, https://doi.org/10.1145/383952.384007 (ACM Special Interest Group on Information Retrieval).
- [15] P. Emerson, Designing an all-inclusive democracy: consensual voting procedures for use in parliaments, councils and committees. Designing an all-Inclusive Democracy: Consensual Voting Procedures for Use in Parliaments, Councils and Committees, Springer Berlin Heidelberg, 2007, https://doi.org/10.1007/978-3-540-33164-3.
- [16] C. Dwork, R. Kumar, M. Naor, D. Sivakumar, Rank aggregation methods for the web, in: Proceedings of the 10th International Conference on World Wide Web, WWW, 2001, pp. 613–622, https://doi.org/10.1145/371920.372165, 2001.
- [17] J. Kemeny, Mathematics Without Numbers, Daedalus, 1959, pp. 571–591. DOI: https://www.jstor.org/stable/20026529.
- [18] S. Lin, J. Ding, Integration of ranked lists via cross entropy Monte Carlo with applications to mRNA and microRNA studies, Biometrics 65 (1) (2009) 9–18, https://doi.org/10.1111/j.1541-0420.2008.01044.x.
- [19] E. Koksalmis, Ö. Kabak, Deriving decision makers' weights in group decision making: an overview of objective methods, Inf. Fusion 49 (2019) 146–160, https://doi.org/10.1016/j.inffus.2018.11.009. May 2018.
- [20] C. Carlsson, R. Fullér, Fuzzy multiple criteria decision making: recent developments, Fuzzy Sets Syst. 78 (2) (1996) 139–153, https://doi.org/10.1016/ 0165-0114(95)00165-4.
- [21] E.U. Choo, B. Schoner, W.C. Wedley, Interpretation of criteria weights in multicriteria decision making, Comput. Ind. Eng. 37 (3) (1999) 527–541, https:// doi.org/10.1016/S0360-8352(00)00019-X.

- [22] T. Bui, M. Jarke, Dss for cooperative multiple criteria group decision making, in: Proceedings of the International Conference on Information Systems, 1984, pp. 101–113.
- [23] Ö. Kabak, B. Ervural, Multiple attribute group decision making: a generic conceptual framework and a classification scheme, Knowl. Based Syst. 123 (2017) 13–30, https://doi.org/10.1016/j.knosys.2017.02.011.
- [24] F. Herrera, E. Herrera-Viedma, J.L. Verdegay, A model of consensus in group decision making under linguistic assessments, Fuzzy Sets Syst. 78 (1) (1996) 73–87, https://doi.org/10.1016/0165-0114(95)00107-7.
- [25] J. Greatorex, T. Dexter, An accessible analytical approach for investigating what happens between the rounds of a Delphi study, J. Adv. Nurs. 32 (4) (2000) 1016–1024, https://doi.org/10.1046/j.1365-2648.2000.t01-1-01569.x.
- [26] C. Okoli, S.D. Pawlowski, The Delphi method as a research tool: an example, design considerations and applications, Inf. Manag. 42 (1) (2004) 15–29, https:// doi.org/10.1016/j.im.2003.11.002.
- [27] Schwenk R., C., & Cosier, R. (1993). Effects of consensus and devil 's advocacy on strategic decision-making. 126–139.
- [28] H.J. Schielke, J.L. Fishman, K. Osatuke, W.B. Stiles, Creative consensus on interpretations of qualitative data: the ward method, Psychother. Res. 19 (4–5) (2009) 558–565, https://doi.org/10.1080/10503300802621180.
- [29] B. Srdjevic, Z. Srdjevic, B. Blagojevic, K. Suvocarev, A two-phase algorithm for consensus building in AHP-group decision making, Appl. Math. Model. 37 (10–11) (2013) 6670–6682, https://doi.org/10.1016/j.apm.2013.01.028.
- [30] M. Akram, K. Zahid, C. Kahraman, Integrated outranking techniques based on spherical fuzzy information for the digitalization of transportation system, Appl. Soft Comput. 134 (2023), https://doi.org/10.1016/j.asoc.2023.109992.
- [31] E. Güner, H. Aygün, Spherical fuzzy soft sets: theory and aggregation operator with its applications, Iran. J. Fuzzy Syst. 19 (2) (2022) 83–97, https://doi.org/ 10.22111/LJFS.2022.6789.
- [32] M. Akram, N. Ramzan, M. Deveci, Linguistic Pythagorean fuzzy CRITIC-EDAS method for multiple-attribute group decision analysis, Eng. Appl. Artif. Intell. 119 (2023), https://doi.org/10.1016/j.engappai.2022.105777.
- [33] Z. Xu, An overview of methods for determining OWA weights, Int. J. Intell. Syst. 20 (8) (2005) 843–865, https://doi.org/10.1002/int.20097.
- [34] R.R. Yager, Connectives and quantifiers in fuzzy sets, Fuzzy Sets Syst. 40 (1) (1991) 39–75, https://doi.org/10.1016/0165-0114(91)90046-S.
- [35] M. Akram, W.A. Dudek, F. Ilyas, Group decision-making based on pythagorean fuzzy TOPSIS method, Int. J. Intell. Syst. 34 (7) (2019) 1455–1475, https://doi. org/10.1002/int.22103.
- [36] H. Garg, G. Kaur, Extended TOPSIS method for multi-criteria group decisionmaking problems under cubic intuitionistic fuzzy environment, Sci. Iran. 27 (1 E) (2020) 396–410, https://doi.org/10.24200/SCI.2018.5307.1194.
- [37] B. Vahdani, S.M. Mousavi, R. Tavakkoli-Moghaddam, Group decision making based on novel fuzzy modified TOPSIS method, Appl. Math. Model. 35 (9) (2011) 4257–4269, https://doi.org/10.1016/j.apm.2011.02.040.
- [38] M. Akram, A. Luqman, J.C.R. Alcantud, An integrated ELECTRE-I approach for risk evaluation with hesitant Pythagorean fuzzy information, Expert Syst. Appl. (2022) 200, https://doi.org/10.1016/j.eswa.2022.116945.
- [39] M. Akram, U. Noreen, M. Deveci, Enhanced ELECTRE II method with 2-tuple linguistic m-polar fuzzy sets for multi-criteria group decision making, Expert Syst. Appl. 213 (2023), https://doi.org/10.1016/j.eswa.2022.119237.
- [40] D. Zindani, S.R. Maity, S. Bhownik, Interval-valued intuitionistic fuzzy TODIM method based on Schweizer–Sklar power aggregation operators and their applications to group decision making, Soft Comput. 24 (18) (2020), https://doi. org/10.1007/s00500-020-04783-1.
- [41] C. Wu, Z. Zhang, W. Zhong, A group decision-making approach based on DST and AHP for new product selection under epistemic uncertainty. Math. Probl. Eng., 2019, https://doi.org/10.1155/2019/4635374, 2019Mcdm.
- [42] L.L. Thurstone, A law of comparative judgment, Psychol. Rev. 34 (4) (1927) 273–286, https://doi.org/10.1037/h0070288.
- [43] Deconde, R.P., Health, P., Division, S., Clegg, N., Biology, H., Hutchinson, F., Knudsen, B., Hutchinson, F., Etzioni, R., Health, P., Division, S., Deconde, R.P., Hawley, S., Falcon, S., Clegg, N., & Knudsen, B. (2006). Combining results of microarray experiments : a rank aggregation approach combining results of microarray experiments : a rank aggregation approach. 5(1), 5–15. 10.2202/ 1544-6115.1204.
- [44] J. Guiver, E. Snelson, Bayesian inference for Plackett-Luce ranking models, in: Proceedings of the 26th International Conference on Machine Learning, ICML, 2009, pp. 377–384, 2009.
- [45] S. Niu, Y. Lan, J. Guo, X. Cheng, Stochastic rank aggregation, in: Proceedings of the 29th Conference - Uncertainty in Artificial Intelligence ,UAI, 2013, pp. 478–487.
- [46] H. Cooper, Synthesizing Research: a Guide for Literature Reviews, 2, Sage, Sage, 1998 (3rd ed.
- [47] Reuters, T. (2017). Web of Science. https://web.archive.org/web/2017022401 3916/http://wokinfo.com/citationconnection/realfacts.
- [48] R.L. Keeney, C.W. Kirkwood, Group decision making using cardinal social welfare functions, Manag. Sci. 22 (4) (1974) 430–437, https://doi.org/10.1287/ mnsc.22.4.430.
- [49] A. Kelemenis, K. Ergazakis, D. Askounis, Support managers' selection using an extension of fuzzy TOPSIS, Expert Syst. Appl. 38 (3) (2011) 2774–2782, https:// doi.org/10.1016/j.eswa.2010.08.068.
- [50] I. Palomares, L. Martínez, A semisupervised multiagent system model to support consensus-reaching processes, IEEE Trans. Fuzzy Syst. 22 (4) (2014) 762–777, https://doi.org/10.1109/TFUZZ.2013.2272588.

D. Boix-Cots et al.

- [52] A. Hafezalkotob, A. Hafezalkotob, A novel approach for combination of individual and group decisions based on fuzzy best-worst method, Appl. Soft Comput. Journal 59 (2017) 316–325, https://doi.org/10.1016/j. asoc.2017.05.036.
- [53] P. Wang, X. Xu, S. Huang, An improved consensus-based model for large group decision making problems considering experts with linguistic weighted information, Group Decis. Negot. 28 (3) (2019) 619–640, https://doi.org/ 10.1007/s10726-019-09615-8.
- [54] Q. Wu, L. Zhou, Y. Chen, H. Chen, An integrated approach to green supplier selection based on the interval type-2 fuzzy best-worst and extended VIKOR methods, Inf. Sci. 502 (2019) 394–417, https://doi.org/10.1016/j. ins.2019.06.049.
- [55] S. Aly, I. Vrana, Evaluating the knowledge, relevance and experience of expert decision makers utilizing the Fuzzy-AHP, Agric. Econ. 54 (11) (2008) 529–535, https://doi.org/10.17221/264-agricecon.
- [56] F. Samanlioglu, Y.E. Taskaya, U.C. Gulen, O. Cokcan, A fuzzy AHP–TOPSIS-based group decision-making approach to IT personnel selection, Int. J. Fuzzy Syst. 20 (5) (2018) 1576–1591, https://doi.org/10.1007/s40815-018-0474-7.
- [57] Z. Chen, S. Wan, J. Dong, An efficiency-based interval type-2 fuzzy multi-criteria group decision making for makeshift hospital selection, Appl. Soft Comput. 115 (2022), https://doi.org/10.1016/j.asoc.2021.108243.
- [58] P. Gupta, M.K. Mehlawat, N. Grover, A Generalized TOPSIS method for intuitionistic Fuzzy multiple attribute group decision making considering different scenarios of attributes weight information, Int. J. Fuzzy Syst. 21 (2) (2019) 369–387, https://doi.org/10.1007/s40815-018-0563-7.
- [59] C. Yang, Q. Wang, W. Peng, J. Zhu, A multi-criteria group decision-making approach based on improved BWM and MULTIMOORA with normal wiggly hesitant fuzzy information, Int. J. Comput. Intell. Syst. 13 (1) (2020) 366–381, https://doi.org/10.2991/ijcis.d.200325.001.
- [60] Y. Sun, H.F. Huang, Algorithm of adjusting weights of decision-makers and attribute in multi-attribute group decision-making, in: Proceedings of the 25th Chinese Control and Decision Conference 2013, CCDC, 2013, pp. 4144–4148, https://doi.org/10.1109/CCDC.2013.6561677.
- [61] S.E. Bodily, Note—a delegation process for combining individual utility functions, Manag. Sci. 25 (10) (1979) 1035–1041, https://doi.org/10.1287/ mnsc.25.10.1035.
- [62] R. Ramanathan, L.S. Ganesh, Group preference aggregation methods employed in AHP: an evaluation and an intrinsic process for deriving members' weightages, Eur. J. Oper. Res. 79 (2) (1994) 249–265, https://doi.org/10.1016/0377-2217 (94)90356-5.
- [63] R.C. van den Honert, Decisional power in group decision making: a note on the allocation of group members' Weights in the Multiplicative AHP and SMART, Group Decis. Negot. 10 (3) (2001) 275–286, https://doi.org/10.1023/A: 1011201501379.
- [64] X. Chen, L. Zhao, H. Liang, A novel multi-attribute group decision-making method based on the MULTIMOORA with linguistic evaluations, Soft Comput. 22 (16) (2018) 5347–5361, https://doi.org/10.1007/s00500-018-3030-3.
- [65] J. Wu, F. Chiclana, E. Herrera-Viedma, Trust based consensus model for social network in an incomplete linguistic information context, Appl. Soft Comput. J. 35 (2015) 827–839, https://doi.org/10.1016/j.asoc.2015.02.023.
- [66] X. Liu, Y. Xu, R. Montes, F. Herrera, Social network group decision making: managing self-confidence-based consensus model with the dynamic importance degree of experts and trust-based feedback mechanism, Inf. Sci. 505 (2019) 215–232, https://doi.org/10.1016/j.ins.2019.07.050.
- [67] Z.P. Tian, R. Nie, J. Wang, Social network analysis-based consensus-supporting framework for large-scale group decision-making with incomplete interval type-2 fuzzy information, Inf. Sci. 502 (2019) 446–471, https://doi.org/10.1016/j. ins.2019.06.053.
- [68] Y. Dong, H. Zhang, E. Herrera-Viedma, Integrating experts' weights generated dynamically into the consensus reaching process and its applications in managing non-cooperative behaviors, Decis. Support Syst. 84 (2016) 1–15, https://doi.org/ 10.1016/j.dss.2016.01.002.
- [69] H.W. Brock, Problem of "utility weights" in group preference aggregation, Oper Res. 28 (1) (1980) 176–187, https://doi.org/10.1287/opre.28.1.176.
- [70] E. Herowati, U. Ciptomulyono, J. Parung, Expertise-based experts importance weights in adverse judgment, ARPN J. Eng. Appl. Sci. 9 (9) (2014) 1428–1435.
- [71] E. Herowati, U. Ciptomulyono, J. Parung, Expertise-based ranking of experts: an assessment level approach, Fuzzy Sets Syst. 315 (2017) 44–56, https://doi.org/ 10.1016/j.fss.2016.09.016.
- [72] D. Cheng, Z. Zhou, F. Cheng, J. Wang, Deriving heterogeneous experts weights from incomplete linguistic preference relations based on uninorm consistency, Knowl. Based Syst. 150 (2018) 150–165, https://doi.org/10.1016/j. knosvs.2018.03.013.
- [73] S. Chakhar, A. Ishizaka, A. Labib, I. Saad, Dominance-based rough set approach for group decisions, Eur. J. Oper. Res. 251 (1) (2016) 206–224, https://doi.org/ 10.1016/j.ejor.2015.10.060.
- [74] F. Chunhua, H. Shi, B. Guozhen, A group decision making method for sustainable design using intuitionistic fuzzy preference relations in the conceptual design stage, J. Clean. Prod. 243 (2020), 118640, https://doi.org/10.1016/j. jclepro.2019.118640.
- [75] I. Ivlev, P. Kneppo, M. Barták, Method for selecting expert groups and determining the importance of experts' judgments for the purpose of managerial

decision-making tasks in health system, E a M: Ekon. Manag. 18 (2) (2015) 57-72, https://doi.org/10.15240/tul/001/2015-2-005.

- [76] H.C. Liu, L.E. Wang, Z. Li, Y.P. Hu, Improving risk evaluation in FMEA with cloud model and hierarchical TOPSIS method, IEEE Trans. Fuzzy Syst. 27 (1) (2019) 84–95, https://doi.org/10.1109/TFUZZ.2018.2861719.
- [77] D. Borissova, A group decision making model considering experts competency: an application in personnel selection, C. R. Acad. Bulg. Des Sci. 71 (11) (2018) 1520–1527, https://doi.org/10.7546/CRABS.2018.11.11.
- [78] M.M.G. Elbarkouky, A.R. Fayek, Fuzzy similarity consensus model for early alignment of construction project teams on the extent of their roles and responsibilities, J. Constr. Eng. Manag. 137 (6) (2011) 432–440, https://doi.org/ 10.1061/(asce)co.1943-7862.0000310.
- [79] H. Sellak, B. Ouhbi, B. Frikh, B. Ikken, Expertise-based consensus building for MCGDM with hesitant fuzzy linguistic information, Inf. Fusion 50 (2019) 54–70, https://doi.org/10.1016/j.inffus.2018.10.003. June 2018.
- [80] Y.T. Bai, B.H. Zhang, X.Y. Wang, X.B. Jin, J.P. Xu, Z.Y. Wang, Expert decision support technique for algal bloom governance in urban lakes based on text analysis, Water 9 (5) (2017), https://doi.org/10.3390/w9050308 (Switzerland).
- [81] S. Chatterjee, M. Bhattacharyya, A probabilistic approach to group decision making, in: Proceedings of the Conference on Human Factors in Computing Systems, 2017, pp. 2445–2451, https://doi.org/10.1145/3027063.3053226. -Proceedings, Part F1276.
- [82] D.P. Slevin, L.W. Boone, E.M. Russo, R.S. Allen, CONFIDE: a collective decisionmaking procedure using confidence estimates of individual judgements, Group Decis. Negot. 7 (2) (1998) 179–194, https://doi.org/10.1023/A:1008650524782.
- [83] H. Liao, R. Tan, M. Tang, An overlap graph model for large-scale group decision making with social trust information considering the multiple roles of experts, Expert Syst. (2020) 1–20, https://doi.org/10.1111/exsy.12659. May.
- [84] von Neumann, J., & Morgenstern, O. (1944). Theory of games and economic behavior,.
- [85] L.S. Shapley, M. Shubik, A method for evaluating the distribution of power in a committee system, Am. Political Sci. Rev. 48 (3) (1954) 787–792, https://doi. org/10.2307/1951053.
- [86] J.F. Banzhaf, Weighted voting doesn't work: a mathematical analysis, Rutgers Law Rev. 19 (2) (1965) 317–343.
- [87] J. Deegan, E.W. Packel, A new index of power for simple n-person games, Int. J. Game Theory 7 (2) (1978) 113–123, https://doi.org/10.1007/BF01753239.
- [88] D.S. Felsenthal, A well-behaved index of a priori P-power for simple N-person games, Homo Oecon. 33 (4) (2016) 367–381, https://doi.org/10.1007/s41412-016-0031-2.
- [89] S. Wan, X. Cheng, C. Chen, J. Dong, L-R geometric consistency definition of triangular multiplicative preference relation in group decision making, Fuzzy Sets Syst. 409 (2021) 85–113, https://doi.org/10.1016/j.fss.2020.07.006.
- [90] H. Theil, On the symmetry approach to the committee decision problem, Manag. Sci. 9 (3) (1963) 380–393, https://doi.org/10.1287/mnsc.9.3.380.
- [91] Z. Xu, W. Zhou, Consensus building with a group of decision makers under the hesitant probabilistic fuzzy environment, Fuzzy Optim. Decis. Mak. 16 (4) (2017) 481–503, https://doi.org/10.1007/s10700-016-9257-5.
- [92] S. Zeng, D. Palacios-Marqués, F. Zhu, A New model for interactive group decision making with intuitionistic fuzzy preference relations, Informatica 27 (4) (2016) 911–928, https://doi.org/10.15388/Informatica.2016.117 (Netherlands).
- [93] N.T. Thong, L.T.H. Lan, S.Y. Chou, L.H. Son, D.D. Dong, T.T. Ngan, An extended TOPSIS method with unknown weight information in dynamic neutrosophic environment, Mathematics 8 (3) (2020) 1–15, https://doi.org/10.3390/ math8030401.
- [94] Z. Wu, J. Xu, X. Jiang, L. Zhong, Two MAGDM models based on hesitant fuzzy linguistic term sets with possibility distributions: VIKOR and TOPSIS, Inf. Sci. 473 (2019) 101–120, https://doi.org/10.1016/j.ins.2018.09.038.
- [95] W. Lu, C. Liang, Y. Ding, A method for determining the objective weights of experts based on evidence similarity in group decision-making, in: Proceedings of the 2008 International Conference on Wireless Communications, Networking and Mobile Computing, WiCOM, 2008, pp. 1–4, https://doi.org/10.1109/ WiCom.2008.2764. 2008.
- [96] X. Wang, Y. Geng, P. Yao, M. Yang, Multiple attribute group decision making approach based on extended VIKOR and linguistic neutrosophic Set, J. Intell. Fuzzy Syst. 36 (1) (2019) 149–160, https://doi.org/10.3233/JIFS-181066.
- [97] P. Gupta, M.K. Mehlawat, N. Grover, Intuitionistic fuzzy multi-attribute group decision-making with an application to plant location selection based on a new extended VIKOR method, Inf. Sci. 370–371 (01) (2016) 184–203, https://doi. org/10.1016/j.ins.2016.07.058.
- [98] S.P. Wan, Q.Y. Wang, J.Y. Dong, The extended VIKOR method for multi-attribute group decision making with triangular intuitionistic fuzzy numbers, Knowl. Based Syst. 52 (2013) 65–77, https://doi.org/10.1016/j.knosys.2013.06.019.
- [99] F. Voorbraak, A computationally efficient approximation of Dempster-Shafer theory, Int. J. Man Mach. Stud. 30 (5) (1989) 525–536, https://doi.org/10.1016/ S0020-7373(89)80032-X.
- [100] A. Fahmi, F. Amin, S. Abdullah, M. Aslam, N. Ul Amin, Cubic Fuzzy multiattribute group decision-making with an application to plant location selected based on a new extended Vikor method, J. Intell. Fuzzy Syst. 37 (1) (2019) 583–596, https://doi.org/10.3233/JIFS-171049.
- [101] R. Krishankumar, Y. Gowtham, I. Ahmed, K.S. Ravichandran, S. Kar, Solving green supplier selection problem using q-rung orthopair fuzzy-based decision framework with unknown weight information, Appl. Soft Comput. J. 94 (2020), 106431, https://doi.org/10.1016/j.asoc.2020.106431.
- [102] R. Krishankumar, K.S. Ravichandran, H. Liao, S. Kar, An integrated decision framework for group decision-making with double hierarchy hesitant fuzzy

linguistic information and unknown weights, Int. J. Comput. Intell. Syst. 13 (1) (2020) 624–637, https://doi.org/10.2991/ijcis.d.200527.002.

- [103] F. Meng, S.M. Chen, R. Yuan, Group decision making with heterogeneous intuitionistic fuzzy preference relations, Inf. Sci. 523 (2020) 197–219, https:// doi.org/10.1016/j.ins.2020.03.010.
- [104] F. Wang, J. Mao, Approach to multicriteria group decision making with Znumbers based on TOPSIS and power aggregation operators, Math. Probl. Eng. (2019), https://doi.org/10.1155/2019/3014387, 2019.
- [105] X.Y. Zhang, J.Q. Wang, Consensus-based framework to MCGDM under multigranular uncertain linguistic environment, J. Intell. Fuzzy Syst. 33 (2) (2017) 1263–1274, https://doi.org/10.3233/JIFS-17202.
- [106] S. Wan, F. Wang, J. Dong, A group decision-making method considering both the group consensus and multiplicative consistency of interval-valued intuitionistic fuzzy preference relations, Inf. Sci. 466 (2018) 109–128, https://doi.org/ 10.1016/j.ins.2018.07.031.
- [107] Y. Xu, P. Gao, L. Martínez, An interindividual iterative consensus model for fuzzy preference relations, Int. J. Intell. Syst. 34 (8) (2019) 1864–1888, https://doi.org/ 10.1002/int.22122.
- [108] S. Wan, W.C. Zou, J.Y. Dong, L. Martínez, A consensual method for multi-criteria group decision-making with linguistic intuitionistic information, Inf. Sci. 582 (2022) 797–832, https://doi.org/10.1016/j.ins.2021.10.030.
- [109] P. Liu, P. Liu, P. Wang, B. Zhu, An extended multiple attribute group decision making method based on q-Rung orthopair fuzzy numbers, IEEE Access 7 (2019) 162050–162061, https://doi.org/10.1109/ACCESS.2019.2951357.
- [110] S. Wan, W.C. Zou, J.Y. Dong, L. Martínez, A probabilistic linguistic dominance score method considering individual semantics and psychological behavior of decision makers, Expert Syst. Appl. 184 (2021), https://doi.org/10.1016/j. eswa.2021.115372.
- [111] P. Zhang, Z. Tao, J. Liu, F. Jin, J. Zhang, An ELECTRE TRI-based outranking approach for multi-attribute group decision making with picture fuzzy sets, J. Intell. Fuzzy Syst. 38 (4) (2020) 4855–4868, https://doi.org/10.3233/jifs-191540.
- [112] P. Liu, S. Cheng, An improved MABAC group decision-making method using regret theory and likelihood in probability multi-valued neutrosophic sets, Int. J. Inf. Technol. Decis. Mak. 19 (5) (2020) 1353–1387, https://doi.org/10.1142/ S0219622020500303.
- [113] W. Chen, Y. Zou, Group decision making under generalized fuzzy soft sets and limited cognition of decision makers, Eng. Appl. Artif. Intell. 87 (2020), 103344, https://doi.org/10.1016/j.engappai.2019.103344. November 2019.
- [114] P. Gupta, M.K. Mehlawat, N. Grover, W. Pedrycz, Multi-attribute group decision making based on extended TOPSIS method under interval-valued intuitionistic fuzzy environment, Appl. Soft Comput. J. 69 (2018) 554–567, https://doi.org/ 10.1016/j.asoc.2018.04.032.
- [115] J.R.P. French, A formal theory of social power, Psychol. Rev. 63 (3) (1956) 181–194, https://doi.org/10.1037/h0046123.
- [116] L.G. Pérez, F. Mata, F. Chiclana, G. Kou, E. Herrera-Viedma, Modelling influence in group decision making, Soft Comput. 20 (4) (2016) 1653–1665, https://doi. org/10.1007/s00500-015-2002-0.
- [117] M. Xue, C. Fu, S.L. Yang, Group consensus reaching based on a combination of expert weight and expert reliability, Appl. Math. Comput. 369 (2020), 124902, https://doi.org/10.1016/j.amc.2019.124902.
- [118] C. Fu, J.B. Yang, S.L. Yang, A group evidential reasoning approach based on expert reliability, Eur. J. Oper. Res. 246 (3) (2015) 886–893, https://doi.org/ 10.1016/j.ejor.2015.05.042.
- [119] Q. Zhou, Z. Wu, A.H. Altalhi, F. Herrera, A two-step communication opinion dynamics model with self-persistence and influence index for social networks based on the DeGroot model, Inf. Sci. 519 (2020) 363–381, https://doi.org/ 10.1016/j.ins.2020.01.052.
- [120] X. Tong, Z.J. Wang, A group decision framework with intuitionistic preference relations and its application to low carbon supplier selection, Int. J. Environ. Res. Public Health (9) (2016) 13, https://doi.org/10.3390/ijerph13090923.
- [121] W. Zhang, Y. Xu, H. Wang, A consensus reaching model for 2-tuple linguistic multiple attribute group decision making with incomplete weight information, Int. J. Syst. Sci. 47 (2) (2016) 389–405, https://doi.org/10.1080/ 00207721.2015.1074761.
- [122] Z.S. Chen, X. Zhang, K. Govindan, X.J. Wang, K.S. Chin, Third-party reverse logistics provider selection: a computational semantic analysis-based multiperspective multi-attribute decision-making approach, Expert Syst. Appl. 166 (2021), 114051, https://doi.org/10.1016/j.eswa.2020.114051.
- [123] X. Li, X. Chen, Extension of the TOPSIS method based on prospect theory and trapezoidal intuitionistic fuzzy numbers for group decision making, J. Syst. Sci. Syst. Eng. 23 (2) (2014) 231–247, https://doi.org/10.1007/s11518-014-5244-y.
- [124] M. Lin, Q. Zhan, Z. Xu, R. Chen, Group decision-making model with hesitant multiplicative preference relations based on regression method and feedback mechanism, IEEE Access 6 (2018) 61130–61150, https://doi.org/10.1109/ ACCESS.2018.2875400.
- [125] J. Ye, Multiple attribute group decision-making methods with unknown weights in intuitionistic fuzzy setting and interval-valued intuitionistic fuzzy setting, Int. J. Gen. Syst. 42 (5) (2013) 489–502, https://doi.org/10.1080/ 03081079.2013.775127.

- [126] H. Gitinavard, H. Ghaderi, M.S. Pishvaee, Green supplier evaluation in manufacturing systems: a novel interval-valued hesitant fuzzy group outranking approach, Soft Comput. 22 (19) (2018) 6441–6460, https://doi.org/10.1007/ s00500-017-2697-1.
- [127] J. Chai, J.N.K. Liu, Z. Xu, A rule-based group decision model for warehouse evaluation under interval-valued Intuitionistic fuzzy environments, Expert Syst. Appl. 40 (6) (2013) 1959–1970, https://doi.org/10.1016/j.eswa.2012.10.003.
- [128] J. Ye, Multiple attribute group decision-making method with completely unknown weights based on similarity measures under single valued neutrosophic environment, J. Intell. Fuzzy Syst. 27 (6) (2014) 2927–2935, https://doi.org/ 10.3233/IFS-141252.
- [129] Q. Qin, F. Liang, L. Li, Y.W. Chen, G.F. Yu, A TODIM-based multi-criteria group decision making with triangular intuitionistic fuzzy numbers, Appl. Soft Comput. J. 55 (2017) 93–107, https://doi.org/10.1016/j.asoc.2017.01.041.
- [130] G. Zhang, J.qiang Wang, T.li. Wang, Multi-criteria group decision-making method based on TODIM with probabilistic interval-valued hesitant fuzzy information, Expert Syst. 36 (4) (2019) 1–27, https://doi.org/10.1111/exsy.12424.
- [131] C. Wei, J. Wu, TODIM method for probabilistic linguistic multiple attribute group decision making based on the similarity measures and entropy, J. Intell. Fuzzy Syst. 37 (5) (2019) 7025–7037, https://doi.org/10.3233/JIFS-191164.
- [132] H.S. Shih, H.J. Shyur, E.S. Lee, An extension of TOPSIS for group decision making, Math. Comput. Model. 45 (7–8) (2007) 801–813, https://doi.org/10.1016/j. mcm.2006.03.023.
- [133] Z. Yue, Deriving decision maker's weights based on distance measure for intervalvalued intuitionistic fuzzy group decision making, Expert Syst. Appl. 38 (9) (2011) 11665–11670, https://doi.org/10.1016/j.eswa.2011.03.046.
- [134] Z. Yue, An extended TOPSIS for determining weights of decision makers with interval numbers, Knowl. Based Syst. 24 (1) (2011) 146–153, https://doi.org/ 10.1016/j.knosys.2010.07.014.
- [135] Z. Yue, Developing a straightforward approach for group decision making based on determining weights of decision makers, Appl. Math. Model. 36 (9) (2012) 4106–4117, https://doi.org/10.1016/j.apm.2011.11.041.
- [136] Z. Yue, Extension of TOPSIS to determine weight of decision maker for group decision making problems with uncertain information, Expert Syst. Appl. 39 (7) (2012) 6343–6350, https://doi.org/10.1016/j.eswa.2011.12.016.
- [137] Z. Yue, TOPSIS-based group decision-making methodology in intuitionistic fuzzy setting, Inf. Sci. 277 (2014) 141–153, https://doi.org/10.1016/j.ins.2014.02.013.
- [138] B. Wu, L. Zong, X. Yan, C. Guedes Soares, Incorporating evidential reasoning and TOPSIS into group decision-making under uncertainty for handling ship without command, Ocean Eng. 164 (2018) 590–603, https://doi.org/10.1016/j. oceaneng.2018.06.054. April.
- [139] Y. Zarook, N. Javadian, J. Rezaeian, A hybrid method based on AHP and TOPSIS with uncertain data for MAGDM problem, Int. J. Ind. Syst. Eng. 19 (1) (2015) 123–136, https://doi.org/10.1504/IJISE.2015.065956.
- [140] D. Kacprzak, A doubly extended TOPSIS method for group decision making based on ordered fuzzy numbers, Expert Syst. Appl. 116 (2019) 243–254, https://doi. org/10.1016/j.eswa.2018.09.023.
- [141] A. Biswas, B. Sarkar, Pythagorean fuzzy TOPSIS for multicriteria group decisionmaking with unknown weight information through entropy measure, Int. J. Intell. Syst. 34 (6) (2019) 1108–1128, https://doi.org/10.1002/int.22088.
- [142] S. Liu, F.T.S. Chan, W. Ran, Multi-attribute group decision-making with multigranularity linguistic assessment information: an improved approach based on deviation and TOPSIS, Appl. Math. Model. 37 (24) (2013) 10129–10140, https:// doi.org/10.1016/j.apm.2013.05.051.
- [143] S.P. Wan, F. Wang, L.L. Lin, J.Y. Dong, An intuitionistic fuzzy linear programming method for logistics outsourcing provider selection, Knowl. Based Syst. 82 (2015) 80–94, https://doi.org/10.1016/j.knosys.2015.02.027.
- [144] Y. Dorfeshan, S. Meysam Mousavi, A group TOPSIS-COPRAS methodology with Pythagorean fuzzy sets considering weights of experts for project critical path problem, J. Intell. Fuzzy Syst. 36 (2) (2019) 1375–1387, https://doi.org/ 10.3233/JIFS-172252.
- [145] H. Gitinavard, S.M. Mousavi, B. Vahdani, A. Siadat, A distance-based decision model in interval-valued hesitant fuzzy setting for industrial selection problems, Sci. Iran. 23 (4) (2016) 1928–1940, https://doi.org/10.24200/sci.2016.3938.
- [146] S. Saffarzadeh, A. Hadi-Vencheh, A. Jamshidi, Weight determination and ranking priority in interval group MCDM, Sci. Iran. 0 (0) (2019), https://doi.org/ 10.24200/sci.2019.51133.2022, 0–0.
- [147] Q. Yang, P.A. Du, Y. Wang, B. Liang, A rough set approach for determining weights of decision makers in group decision making, PLoS ONE 12 (2) (2017) 1–16, https://doi.org/10.1371/journal.pone.0172679.
- [148] Z. Jiang, Y. Wang, Multiattribute group decision making with unknown decision expert weights information in the framework of interval intuitionistic trapezoidal fuzzy numbers, Math. Probl. Eng. (2014), https://doi.org/10.1155/2014/ 635476, 2014.
- [149] H. Hamdani, R. Wardoyo, K. Mustofa, A method of weight update in group decision-making to accommodate the interests of all the decision makers, Int. J. Intell. Syst. Appl. 9 (8) (2017) 1–10, https://doi.org/10.5815/ijisa.2017.08.01.
- [150] S. Wan, L. Zhong, J. Dong, A new method for group decision making with hesitant fuzzy preference relations based on multiplicative consistency, IEEE Trans. Fuzzy Syst. 28 (7) (2020) 1449–1463, https://doi.org/10.1109/TFUZZ.2019.2914008.

- [151] E. Aydoğdu, E. Güner, B. Aldemir, H. Aygün, Complex spherical fuzzy TOPSIS based on entropy, Expert Syst. Appl. 215 (2023), https://doi.org/10.1016/j. eswa.2022.119331.
- [152] Z. Yue, Approach to group decision making based on determining the weights of experts by using projection method, Appl. Math. Model. 36 (7) (2012) 2900–2910, https://doi.org/10.1016/j.apm.2011.09.068.
- [153] Z. Yue, An intuitionistic fuzzy projection-based approach for partner selection, Appl. Math. Model. 37 (23) (2013) 9538–9551, https://doi.org/10.1016/j. apm.2013.05.007.
- [154] Q. Yang, P. Du, A straightforward approach for determining the weights of decision makers based on angle cosine and projection method, Int. J. Ind. Manuf. Eng. 9 (10) (2015) 3328–3334.
- [155] G.Ii Xu, F Liu, An approach to group decision making based on interval multiplicative and fuzzy preference relations by using projection, Appl. Math. Model. 37 (6) (2013) 3929–3943, https://doi.org/10.1016/j.apm.2012.08.007
- [156] G. Sun, A group decision making method based on projection method and score function under IVIFS environment, Br. J. Math. Comput. Sci. 9 (1) (2015) 62–72, https://doi.org/10.9734/bjmcs/2015/9549.
- [157] H. Liao, C. Zhang, L. Luo, A multiple attribute group decision making method based on two novel intuitionistic multiplicative distance measures, Inf. Sci. 467 (2018) 766–783, https://doi.org/10.1016/j.ins.2018.05.023.
- [158] L. Luo, C. Zhang, H. Liao, Distance-based intuitionistic multiplicative MULTIMOORA method integrating a novel weight-determining method for multiple criteria group decision making, Comput. Ind. Eng. 131 (2019) 82–98, https://doi.org/10.1016/j.cie.2019.03.038. July 2018.
- [159] D. Boix-Cots, F. Pardo-Bosch, P. Pujadas, A hierarchical integration method under social constraints to maximize satisfaction in multiple criteria group decision making systems, Expert Syst. Appl. 216 (2023), 119471, https://doi.org/ 10.1016/j.eswa.2022.119471.
- [160] C. Gong, W. Li, P. Yi, Rank-based analysis method to determine OWA weights and its application in group decision making, Int. J. Intell. Syst. 34 (7) (2019) 1685–1699, https://doi.org/10.1002/int.22116.
- [161] Z. Li, Q. Zhang, X. Du, X. Qian, B. Lev, A hesitant fuzzy multi-criteria group decision making method for college applicants' learning potential evaluation, J. Data Inf. Manag. 1 (1–2) (2019) 65–75, https://doi.org/10.1007/s42488-019-00006-x.
- [162] Y. Lin, Y. Wang, Group decision making with consistency of intuitionistic fuzzy preference relations under uncertainty, IEEE/CAA J. Autom. Sin. 5 (3) (2018) 741–748, https://doi.org/10.1109/JAS.2016.7510037.
- [163] Z. Ma, J. Zhu, Y. Chen, A probabilistic linguistic group decision-making method from a reliability perspective based on evidential reasoning, IEEE Trans. Syst. Man Cybern. Syst. 50 (7) (2020) 2421–2435, https://doi.org/10.1109/ TSMC.2018.2815716.
- [164] J. Xu, Z. Wu, A maximizing consensus approach for alternative selection based on uncertain linguistic preference relations, Comput. Ind. Eng. 64 (4) (2013) 999–1008, https://doi.org/10.1016/j.cie.2013.01.009.
- [165] Z. Xu, X. Cai, Minimizing group discordance optimization model for deriving expert weights, Group Decis. Negot. 21 (6) (2012) 863–875, https://doi.org/ 10.1007/s10726-011-9253-7.
- [166] S. Wan, H. Yuan, J. Dong, Decision making with incomplete interval multiplicative preference relations based on stochastic program and interval category, Inf. Sci. 570 (2021) 403–427, https://doi.org/10.1016/j. ins.2021.03.005.
- [167] S. Abootalebi, A. Hadi-Vencheh, A. Jamshidi, An improvement to determining expert weights in group multiple attribute decision making problem, Group Decis. Negot. 27 (2) (2018) 215–221, https://doi.org/10.1007/s10726-018-9555-0.
- [168] Y. Xu, D. Rui, H. Wang, A dynamically weight adjustment in the consensus reaching process for group decision-making with hesitant fuzzy preference relations, Int. J. Syst. Sci. 48 (6) (2017) 1311–1321, https://doi.org/10.1080/ 00207721.2016.1255803.
- [169] Y. Xu, H. Sun, H. Wang, Optimal consensus models for group decision making under linguistic preference relations, Int. Trans. Oper. Res. 23 (6) (2016) 1201–1228, https://doi.org/10.1111/itor.12154.
- [170] F. Meng, C. Tan, X. Chen, Multiplicative consistency analysis for interval fuzzy preference relations: a comparative study, Omega 68 (932) (2017) 17–38, https://doi.org/10.1016/j.omega.2016.05.006 (United Kingdom).
- [171] W. Li, G. Guo, X. Zhou, Generalizing and integrating TOPSIS and cook-seiford method for multicriteria group decision-making with both cardinal and ordinal data, Math. Probl. Eng. (2015) 13–15, https://doi.org/10.1155/2015/154848, 2015.
- [172] Z. Zhang, C. Guo, A VIKOR-based approach to group decision making with uncertain preference ordinals and incomplete weight information, Informatica 27 (3) (2016) 689–708, https://doi.org/10.15388/Informatica.2016.106 (Netherlands).
- [173] H.M. Regan, M. Colyvan, L. Markovchick-Nicholls, A formal model for consensus and negotiation in environmental management, J. Environ. Manag. 80 (2) (2006) 167–176, https://doi.org/10.1016/j.jenvman.2005.09.004.
- [174] Q. Dong, O. Cooper, A peer-to-peer dynamic adaptive consensus reaching model for the group AHP decision making, Eur. J. Oper. Res. 250 (2) (2016) 521–530, https://doi.org/10.1016/j.ejor.2015.09.016.
- [175] Y. Ji, P. Li, Z. Wu, D. Qu, Reaching consensus based on the opinion dynamics in social networks, Arab. J. Sci. Eng. 46 (2) (2021) 1677–1690, https://doi.org/ 10.1007/s13369-020-04891-y.
- [176] W.C. Zou, S. Wan, S.M. Chen, A fairness-concern-based LINMAP method for heterogeneous multi-criteria group decision making with hesitant fuzzy linguistic

truth degrees, Inf. Sci. 612 (2022) 1206–1225, https://doi.org/10.1016/j. ins.2022.08.111.

- [177] A.H. Liu, S. Wan, J.Y. Dong, An axiomatic design-based mathematical programming method for heterogeneous multi-criteria group decision making with linguistic fuzzy truth degrees, Inf. Sci. 571 (2021) 649–675, https://doi.org/ 10.1016/j.ins.2021.04.091.
- [178] F. Meng, Q. An, X. Chen, A consistency and consensus-based method to group decision making with interval linguistic preference relations, J. Oper. Res. Soc. 67 (11) (2016) 1419–1437, https://doi.org/10.1057/jors.2016.28.
- [179] S.ping Wan, G.li Xu, J.ying Dong, A novel method for group decision making with interval-valued Atanassov intuitionistic fuzzy preference relations, Inf. Sci. 372 (2016), https://doi.org/10.1016/j.ins.2016.08.019.
- [180] S. Wan, J. Yan, J.Y. Dong, Personalized individual semantics based consensus reaching process for large-scale group decision making with probabilistic linguistic preference relations and application to COVID-19 surveillance, Expert Syst. Appl. (2022) 191, https://doi.org/10.1016/j.eswa.2021.116328.
- [181] C. Zhang, H. Liao, L. Luo, Z. Xu, Multiplicative consistency analysis for q-rung orthopair fuzzy preference relation, Int. J. Intell. Syst. 35 (1) (2020) 38–71, https://doi.org/10.1002/int.22197.
- [182] H.S. Lee, Optimal consensus of fuzzy opinions under group decision making environment, Fuzzy Sets Syst. 132 (3) (2002) 303–315, https://doi.org/10.1016/ S0165-0114(02)00056-8.
- [183] J. Li, Z.X. Wang, Deriving priority weights from hesitant fuzzy preference relations in view of additive consistency and consensus, Soft Comput. 23 (24) (2019) 13691–13707, https://doi.org/10.1007/s00500-019-03908-5.
- [184] R.O. Parreiras, P.Y. Ekel, J.S.C. Martini, R.M. Palhares, A flexible consensus scheme for multicriteria group decision making under linguistic assessments, Inf. Sci. 180 (7) (2010) 1075–1089, https://doi.org/10.1016/j.ins.2009.11.046.
- [185] X. Zhang, Z. Xu, Deriving experts' weights based on consistency maximization in intuitionistic fuzzy group decision making, J. Intell. Fuzzy Syst. 27 (1) (2014) 221–233, https://doi.org/10.3233/IFS-130991.
- [186] X. Zhang, Z. Xu, Soft computing based on maximizing consensus and fuzzy TOPSIS approach to interval-valued intuitionistic fuzzy group decision making, Appl. Soft Comput. J. 26 (2015) 42–56, https://doi.org/10.1016/j. asoc.2014.08.073.
- [187] G.L. Xu, A Consensus Reaching Model with Minimum Adjustments in Interval-Valued Intuitionistic MAGDM, Math. Probl. Eng. (2018), https://doi.org/ 10.1155/2018/9070813, 2018.
- [188] Z. Jia, Y. Zhang, Interval-valued intuitionistic fuzzy multiple attribute group decision making with uncertain weights, Math. Probl. Eng. (3) (2019), https:// doi.org/10.1155/2019/5092147, 2019.
- [189] S. Wan, W.B. Huang Cheng, J.Y. Dong, Interactive multi-criteria group decisionmaking with probabilistic linguistic information for emergency assistance of COVID-19, Appl. Soft Comput. 107 (2021), https://doi.org/10.1016/j. asoc.2021.107383.
- [190] L. Yu, K.K. Lai, A distance-based group decision-making methodology for multi-person multi-criteria emergency decision support, Decis Support Syst 51 (2) (2011) 307–315, https://doi.org/10.1016/j.dss.2010.11.024.
 [191] M. Chen, Z. Wan, X. Chen, New Min-Max approach to optimal choice of the
- [191] M. Chen, Z. Wan, X. Chen, New Min-Max approach to optimal choice of the weights in multi-criteria group decision-making problems, Appl. Sci. 5 (4) (2015) 998–1015, https://doi.org/10.3390/app5040998 (Switzerland).
- [192] Y. Li, X. Lian, C. Lu, Z. Wang, A Large group decision making approach based on TOPSIS framework with unknown weights information, in: Proceedings of the MATEC Web of Conferences 100, 2017, pp. 1561–1570, https://doi.org/10.1051/ matecconf/201710002013.
- [193] S. Wan, F. Wang, J. Dong, Additive consistent interval-valued Atanassov intuitionistic fuzzy preference relation and likelihood comparison algorithm based group decision making, Eur J Oper Res 263 (2) (2017) 571–582, https:// doi.org/10.1016/j.ejor.2017.05.022.
- [194] Y. Dong, J. Xiao, H. Zhang, T. Wang, Managing consensus and weights in iterative multiple-attribute group decision making, Appl. Soft Comput. J. 48 (2016) 80–90, https://doi.org/10.1016/j.asoc.2016.06.029.
- [195] J. Pang, J. Liang, P. Song, An adaptive consensus method for multi-attribute group decision making under uncertain linguistic environment, Appl. Soft Comput. J. 58 (2017) 339–353, https://doi.org/10.1016/j.asoc.2017.04.039.
- [196] X.X. Zhang, B.F. Ge, J. Jiang, K.W. Yang, A new consensus model for group decision making using fuzzy linguistic preference relations with heterogeneous experts, J. Intell. Fuzzy Syst. 30 (1) (2016) 171–182, https://doi.org/10.3233/ IFS-151744.
- [197] X. Tan, J. Zhu, Y. Zhang, A consensus reaching process with quantum subjective adjustment in linguistic group decision making, Inf. Sci. 533 (2020) 150–168, https://doi.org/10.1016/j.ins.2020.05.003.
- [198] R.M. Rodríguez, Á. Labella, G.de Tré, L Martínez, A large scale consensus reaching process managing group hesitation, Knowl. Based Syst. 159 (2018) 86–97, https://doi.org/10.1016/j.knosys.2018.06.009. June.
- [199] M. Tang, H. Liao, J. Xu, D. Streimikiene, X. Zheng, Adaptive consensus reaching process with hybrid strategies for large-scale group decision making, Eur J Oper Res 282 (3) (2020) 957–971, https://doi.org/10.1016/j.ejor.2019.10.006.
- [200] Y. Yang, J. He, A novel method based on fixed point iteration and improved topsis method for multi-attribute group decision making, Int. J. Innov. Comput. Inf. Control 17 (1) (2021) 15–29, https://doi.org/10.24507/ijicic.17.01.15.
- [201] D. Ben-Arieh, Z. Chen, Linguistic-labels aggregation and consensus measure for autocratic decision making using group recommendations, IEEE Trans. Syst. Man Cybern. Part A Syst. Hum. 36 (3) (2006) 558–568, https://doi.org/10.1109/ TSMCA.2005.853488.

- [202] W. Wu, G. Kou, Y. Peng, A consensus facilitation model based on experts' weights for investment strategy selection, J. Oper. Res. Soc. 69 (9) (2018) 1435–1444, https://doi.org/10.1080/01605682.2017.1398203.
- [203] F.J. Cabrerizo, R. Heradio, I.J. Pérez, E. Herrera-Viedma, A selection process based on additive consistency to deal with incomplete fuzzy linguistic information, J. Univ. Comput. Sci. 16 (1) (2010) 62–81.
- [204] S. Wan, G.L. Xu, J. Dong, An Atanassov intuitionistic fuzzy programming method for group decision making with interval-valued Atanassov intuitionistic fuzzy preference relations, Appl. Soft Comput. J. 95 (2020), https://doi.org/10.1016/j. asoc.2020.106556.
- [205] A. Toloie-Eshlaghy, E.N. Farokhi, Measuring the importance and the weight of decision makers in the criteria weighting activities of group decision making process, Am. J. Sci. Res. 24 (24) (2011) 6–12.
- [206] W. Wu, G. Kou, Y. Peng, D. Ergu, Improved AHP-group decision making for investment strategy selection, Technol. Econ. Dev. Econ. 18 (2) (2012) 299–316, https://doi.org/10.3846/20294913.2012.680520.
- [207] Z. Xu, X. Cai, Deriving weights from interval multiplicative preference relations in group decision making, Group Decis. Negot. 23 (4) (2014) 695–713, https://doi. org/10.1007/s10726-012-9315-5.
- [208] W. Liu, Z. Lin, F. Wen, G. Ledwich, Analysis and optimisation of the preferences of decision-makers in black-start group decision-making, IET Gener. Transm. Distrib. 7 (1) (2013) 14–23, https://doi.org/10.1049/iet-gtd.2012.0093.
- [209] H. Hashemi, S.M. Mousavi, E.K. Zavadskas, A. Chalekaee, Z. Turskis, A new group decision model based on Grey-Intuitionistic Fuzzy-ELECTRE and VIKOR for contractor assessment problem, Sustainability (5) (2018) 10, https://doi.org/ 10.3390/su10051635 (Switzerland).
- [210] S.L. RazaviToosi, J.M.V. Samani, A fuzzy group decision making framework based on ISM-FANP-FTOPSIS for evaluating watershed management strategies, Water Resour. Manag. 33 (15) (2019) 5169–5190, https://doi.org/10.1007/s11269-019-02423-4.
- [211] P. Wu, J. Liu, L. Zhou, H. Chen, Algorithm for improving additive consistency of linguistic preference relations with an integer optimization model, Appl. Soft Comput. J. 86 (2020), 105955, https://doi.org/10.1016/j.asoc.2019.105955.
- [212] C. Yue, Entropy-based weights on decision makers in group decision-making setting with hybrid preference representations, Appl. Soft Comput. J. 60 (2017) 737–749, https://doi.org/10.1016/j.asoc.2017.07.033.
- [213] J. Ye, Multiple Attribute Group Decision-Making Methods with Completely Unknown Weights in Intuitionistic Fuzzy Setting and Interval-Valued Intuitionistic Fuzzy Setting, Group Decis Negot 22 (2) (2013) 173–188, https:// doi.org/10.1007/s10726-011-9255-5.
- [214] D. Cheng, F. Cheng, Z. Zhou, J. Wang, Group prioritisation with unknown expert weights in incomplete linguistic context, Int. J. Syst. Sci. 48 (12) (2017) 2633–2643, https://doi.org/10.1080/00207721.2017.1337947.
- [215] S. Çalı, Ş.Y. Balaman, A novel outranking based multi criteria group decision making methodology integrating ELECTRE and VIKOR under intuitionistic fuzzy environment, Expert Syst. Appl. 119 (2019) 36–50, https://doi.org/10.1016/j. eswa.2018.10.039.
- [216] Y. Li, Y. Shan, P. Liu, An extended TODIM method for group decision making with the interval intuitionistic fuzzy sets, Math. Probl. Eng. (2015), https://doi.org/ 10.1155/2015/672140, 2015.
- [217] X. Mao, H. Wu, S. Wan, A personalized comprehensive cloud-based method for heterogeneous MAGDM and application in COVID-19, CMES - Comput. Model. Eng. Sci. (1) (2022) 131, https://doi.org/10.32604/cmes.2022.019501.
- [218] L. Jin, R. Mesiar, G. Qian, Weighting models to generate weights and capacities in multicriteria group decision making, IEEE Trans. Fuzzy Syst. 26 (4) (2018) 2225–2236, https://doi.org/10.1109/TFUZZ.2017.2769041.
- [219] H. Liao, R. Qin, C. Gao, X. Wu, A. Hafezalkotob, F. Herrera, Score-HeDLiSF: a score function of hesitant fuzzy linguistic term set based on hesitant degrees and linguistic scale functions: an application to unbalanced hesitant fuzzy linguistic MULTIMOORA, Inf. Fusion 48 (2019) 39–54, https://doi.org/10.1016/j. inffus.2018.08.006. May 2018.
- [220] Y. Wang, H. Chen, L. Zhou, Logarithm compatibility of interval multiplicative preference relations with an application to determining the optimal weights of experts in the group decision making, Group Decis. Negot. 22 (4) (2013) 759–772, https://doi.org/10.1007/s10726-012-9291-9.
- [221] Z. Ma, J. Zhu, K. Ponnambalam, Y. Chen, S. Zhang, Group decision-making with linguistic cognition from a reliability perspective, Cognit. Comput. 11 (2) (2019) 172–192, https://doi.org/10.1007/s12559-019-9622-0.
- [222] X. Zhou, F. Zhang, X. Hui, K. Li, Group decision-making method based on entropy and experts cluster analysis, J. Syst. Eng. Electron. 22 (3) (2011) 468–472, https://doi.org/10.3969/j.issn.1004-4132.2011.03.015.
- [223] X. Qi, C. Liang, J. Zhang, Generalized cross-entropy based group decision making with unknown expert and attribute weights under interval-valued intuitionistic fuzzy environment, Comput. Ind. Eng. 79 (2015) 52–64, https://doi.org/ 10.1016/j.cie.2014.10.017.
- [224] B. Liu, Y. Shen, Y. Chen, X. Chen, Y. Wang, A two-layer weight determination method for complex multi-attribute large-group decision-making experts in a linguistic environment, Inf. Fusion 23 (2015) 156–165, https://doi.org/10.1016/ j.inffus.2014.05.001.
- [225] Z. Zhang, S.M. Chen, A consistency and consensus-based method for group decision making with hesitant fuzzy linguistic preference relations, Inf. Sci. 501 (2019) 317–336, https://doi.org/10.1016/j.ins.2019.05.086.

- [226] Z. Zhang, S.M. Chen, Group decision making based on acceptable multiplicative consistency and consensus of hesitant fuzzy linguistic preference relations, Inf. Sci. 541 (2020) 531–550, https://doi.org/10.1016/j.ins.2020.07.024.
- [227] B. Wang, J. Liang, Y. Qian, Determining decision makers' weights in group ranking: a granular computing method, Int. J. Mach. Learn. Cybern. 6 (3) (2015) 511–521, https://doi.org/10.1007/s13042-014-0278-5.
- [228] S.M. Chen, S.H. Cheng, T.E. Lin, Group decision making systems using group recommendations based on interval fuzzy preference relations and consistency matrices, Inf. Sci. 298 (2015) 555–567, https://doi.org/10.1016/j. ins.2014.11.027 (Nv).
- [229] Z. Zhang, S.M. Chen, C. Wang, Group decision making based on multiplicative consistency and consensus of fuzzy linguistic preference relations, Inf. Sci. 509 (2020) 71–86, https://doi.org/10.1016/j.ins.2019.09.002.
- [230] X.W. Qi, J.L. Zhang, C.Y. Liang, Multiple attributes group decision-making under interval-valued dual hesitant fuzzy unbalanced linguistic environment with prioritized attributes and unknown decision-makers' weights, Information (6) (2018) 9, https://doi.org/10.3390/info9060145 (Switzerland).
- [231] Z. Chen, S. Wan, J. Dong, An integrated interval-valued intuitionistic fuzzy technique for resumption risk assessment amid COVID-19 prevention, Inf. Sci. 619 (2023) 695–721, https://doi.org/10.1016/j.ins.2022.11.028.
- [232] G.L. Xu, S.P. Wan, J.Y. Dong, An Entropy-Based Method for Probabilistic Linguistic Group Decision Making and its Application of Selecting Car Sharing Platforms, Informatica 31 (3) (2020) 621–658, https://doi.org/10.15388/20-INFOR423 (Netherlands).
- [233] J. Pang, X. Guan, J. Liang, B. Wang, P. Song, Multi-attribute group decisionmaking method based on multi-granulation weights and three-way decisions, Int. J. Approx. Reason. 117 (2020) 122–147, https://doi.org/10.1016/j. ijar.2019.11.008.
- [234] B. Liu, T. Huo, X. Wang, Q. Shen, Y. Chen, The decision model of the intuitionistic fuzzy group bid evaluation for urban infrastructure projects considering social costs, Can. J. Civ. Eng. 40 (3) (2013) 263–273, https://doi.org/10.1139/cjce-2012-0283.
- [235] B. Liu, S. Guo, K. Yan, L. Li, X. Wang, Double weight determination method for experts of complex multi-attribute large-group decision-making in interval-valued intuitionistic fuzzy environment, J. Syst. Eng. Electron. 28 (1) (2017) 88–96, https://doi.org/10.21629/JSEE.2017.01.11.
- [236] Y. Li, Y. Cheng, Q. Mou, S. Xian, Novel cross-entropy based on multi-attribute group decision-making with unknown experts' weights under interval-valued intuitionistic fuzzy environment, Int. J. Comput. Intell. Syst. 13 (1) (2020) 1295–1304, https://doi.org/10.2991/ijcis.d.200817.001.
- [237] S. Pramanik, S. Dalapati, S. Alam, F. Smarandache, T.K. Roy, NS-cross entropybased MAGDM under single-valued neutrosophic set environment, Information 9 (2) (2018) 1–21, https://doi.org/10.3390/info9020037 (Switzerland).
- [238] R. Davoudabadi, S.M. Mousavi, V. Mohagheghi, A new last aggregation method of multi-attributes group decision making based on concepts of TODIM, WASPAS and TOPSIS under interval-valued intuitionistic fuzzy uncertainty, Knowl. Inf. Syst. 62 (4) (2020) 1371–1391, https://doi.org/10.1007/s10115-019-01390-x.
- [239] S.P. Wan, G.L. Xu, F. Wang, J.Y. Dong, A new method for Atanassov's intervalvalued intuitionistic fuzzy MAGDM with incomplete attribute weight information, Inf. Sci. 316 (168) (2015) 329–347, https://doi.org/10.1016/j. ins.2015.04.019.
- [240] X. Chen, L. Zhao, M.S. Özdemir, H. Liang, Mixed strategy to allocate resources with air pollution treatment in China: based on the analytic network process and large-group decision-making method, Environ. Sci. Pollut. Res. 25 (17) (2018) 16885–16899, https://doi.org/10.1007/s11356-018-1826-4.
- [241] K. Jabeur, J.M. Martel, S.ben. Khélifa, A distance-based collective preorder integrating the relative importance of the group's members, Group Decision and Negotiation 13 (4) (2004) 327–349, https://doi.org/10.1023/B: GRUP.0000042894.00775.75.
- [242] Q. Yang, Y.L. Li, K.S. Chin, Constructing novel operational laws and information measures for proportional hesitant fuzzy linguistic term sets with extension to PHFL-VIKOR for group decision making, Int. J. Comput. Intell. Syst. 12 (2) (2019) 998–1018, https://doi.org/10.2991/ijcis.d.190902.001.
- [243] S. Liu, F.T.S. Chan, W. Ran, Decision making for the selection of cloud vendor: an improved approach under group decision-making with integrated weights and objective/subjective attributes, Expert Syst. Appl. 55 (2016) 37–47, https://doi. org/10.1016/j.eswa.2016.01.059, 2016.
- [244] V. Mohagheghi, S.M. Mousavi, M. Aghamohagheghi, B. Vahdani, A new approach of multi-criteria analysis for the evaluation and selection of sustainable transport investment projects under uncertainty: a case study, Int. J. Comput. Intell. Syst. 10 (1) (2017) 605–626, https://doi.org/10.2991/ijcis.2017.10.1.41.
- [245] H. Mianabadi, A. Afshar, A new method to evaluate weights of decision makers and its application in water resource management, in: Proceedings of the 13th IWRA World Water Congress, 2008, pp. 1–10, in: http://196.36.166.88/iw ra/Proceedings/Details.aspx?id=5.
- [246] Z. Xu, Group decision making based on multiple types of linguistic preference relations, Inf. Sci. 178 (2) (2008) 452–467, https://doi.org/10.1016/j. ins.2007.05.018.
- [247] A.I. Ölçer, A.Y. Odabaşi, A new fuzzy multiple attributive group decision making methodology and its application to propulsion/manoeuvring system selection problem, Eur. J. Oper. Res. 166 (1) (2005) 93–114, https://doi.org/10.1016/j. ejor.2004.02.010. SPEC. ISS.

- [248] S. Wan, J. Yan, J.Y. Dong, Trust and personalized individual semantics based fusion method for heterogeneous multi-criteria group decision making and application to live streaming commerce, Expert Syst. Appl. 208 (2022), https:// doi.org/10.1016/j.eswa.2022.118151.
- [249] L. Bai, K. Zhang, H. Shi, M. An, X. Han, Project portfolio resource risk assessment considering project interdependency by the fuzzy Bayesian network, Complexity (2020), https://doi.org/10.1155/2020/5410978, 2020.
- [250] W. Liu, L. Li, An approach to determining the integrated weights of decision makers based on interval number group decision matrices, Knowl. Based Syst. 90 (2015) 92–98, https://doi.org/10.1016/j.knosys.2015.09.029.
- [251] L. Chen, N. Luo, X. Gou, A novel q-rung orthopair fuzzy TODIM approach for multi-criteria group decision making based on Shapley value and relative

entropy, J. Intell. Fuzzy Syst. 40 (1) (2021) 235–250, https://doi.org/10.3233/ JIFS-191374.

- [252] X. Liu, Y. Xu, Y. Ge, W. Zhang, F. Herrera, A group decision making approach considering self-confidence behaviors and its application in environmental pollution emergency management, Int. J. Environ. Res. Public Health (3) (2019) 16, https://doi.org/10.3390/ijerph16030385.
- [253] Z. Ren, Z. Xu, H. Wang, Multi-criteria group decision-making based on quasiorder for dual hesitant fuzzy sets and professional degrees of decision makers, Appl. Soft Comput. J. 71 (2018) 20–35, https://doi.org/10.1016/j. asoc.2018.06.027.